\documentclass[10pt]{studiamnew}
\usepackage{graphicx}
\usepackage{amsmath}
\usepackage{amssymb}
\sloppy

\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{algorithm}[theorem]{Algorithm}
\newtheorem{assumption}[theorem]{Assumption}
\newtheorem{axiom}[theorem]{Axiom}
\newtheorem{case}[theorem]{Case}
\newtheorem{condition}[theorem]{Condition}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{criterion}[theorem]{Criterion}
\newtheorem{definition}[theorem]{Definition}

\theoremstyle{definition}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{example}[theorem]{Example}
\newtheorem{exercise}[theorem]{Exercise}
\newtheorem{solution}[theorem]{Solution}
\newtheorem{notation}[theorem]{Notation}

\renewcommand{\theequation}{\thesection.\arabic{equation}}
\numberwithin{equation}{section}

\begin{document}
\begin{thebibliography}{99}


\bibitem{alakoya}  Alakoya, T.O.,  Mewomo, O.T.,  \emph{Viscosity S-Iteration Method with Inertial Technique and Self-Adaptive Step Size for Split Variational Inclusion, Equilibrium and Fixed Point Problems},   Comput. Appl. Math.,  {\bf 41} (1) (2022), Paper No. 39, 31 pp.
\bibitem{alakoya1}  Alakoya, T.O.,  Mewomo, O.T., \emph{S-Iteration inertial subgradient extragradient method for variational inequality and fixed point problems},
 Optimization, (2023) DOI: 10.1080/02331934.2023.2168482.
\bibitem{alakoya1}   Alakoya, T.O.,  Uzor, V.A., Mewomo, O.T.,  Yao, J.-C., \emph{On a system of monotone variational inclusion problems with fixed-point constraint}, J. Inequ. Appl.,  \textbf{2022} (2022), Paper No. 47, 33 pp.
\bibitem{alakoya2}  Alakoya, T.O., Uzor, V.A., Mewomo, O.T., \emph{A new projection and contraction method for solving split monotone variational inclusion, pseudomonotone variational inequality, and common fixed point problems},  Comput. Appl. Math., {\bf 42} (2023), Art. No. 33 pp.
			
\bibitem{alv}   Alvarez, F., Attouch, H., \emph{An inertial proximal method for monotone operators via discretization of a nonlinera oscillator with damping},  Set-valued Anal., \textbf{9} (2001), 3-11.
		
	
\bibitem{atto}    Attouch, H., Peypouquet, J., Redont, P., \emph{Backward-forward algorithm for structured monotone inclusions in Hilbert spaces},  J. Math. Anal. Appl., \textbf{457} (2018), 1095-1117.
		
\bibitem{bau}   Bauschke, H.H.,  Combettes, P.L., \emph{Convex Analysis and Monotone Operator Theory in Hilbert Spaces}, CMS Books in Mathematics, Springer, New York (2011), Vol (408).
		
\bibitem{BlOe}   Blum, E., Oettli, W., \emph{From optimization and variational inequalities to equilibrium problems}, The Mathematics Student, \textbf{63} (1-4) (1994), 123-145.
		
\bibitem{bot}  Bot, R.I., Csetnek, E.R., \emph{An inertial forward-backward-forward primal-dual splitting algorithm for solving monotone inclusion problems},  Numer. Algorithms, \textbf{71} (2016), 519-540.
		
\bibitem{brez}   Brézis, H., Chapitre II (1973) \emph{Operateurs maximaux monotones},  North-Holland Math. Stud., {\bf 5}, 19–51
		
\bibitem{CeYa}    Ceng, L.C., Yao, J.C., \emph{A hybrid iterative scheme for mixed equilibrium problems and fixed point problems},  J. Comput. Appl. Math., \textbf{214} (2008), 186-201.
		
\bibitem{CeGiRe}  Censor, Y., Gibali, A., Reich, S., \emph{Algorithms for the split variational inequality problem},  Numer. Algorithms, \textbf{59} (2012), 301-323.
		
\bibitem{ChYa}   Chen, R., Yao, Y., \emph{Strong convergence theorems for strict pseudo-contractions in Hilbert spaces},  J. Appl. Math. Comput., \textbf{32} (2010), 69-82.
		
		\bibitem{cho} Cholamjiak, W., Cholamjiak, P., Suantai, S., \emph{An inertial forward-backward splitting method for solving inclusion problems in
		Hilbert spaces},  J. Fixed Point Theory Appl., \textbf{20}(1)(2018), 1-17.
		
			
		\bibitem{comb}   Combettes, P.L.,  Wajs, V.R., \emph{Signal recovery by proximal forward-backward splitting},  Multiscale Model Simul., \textbf{4} (2005), 1168-1200.
	\bibitem{dph}  Deepho, J., Martinez-Moreno, J., Kumam, P.,  \emph{A viscosity of Cesàro mean approximation method for split generalized equilibrium, variational inequality and fixed point problems},  J. Nonlinear Sci. Appl., \textbf{9} (4) (2016), 1475 – 1496.
						
		\bibitem{eck1}  Ecksten, J., Svaiter, B.F., \emph{A family of projective splitting splitting methods for the sum of two maximal monotone operators},  Math. Progr. Ser B, \textbf{111} (2008), 173-199.
		
		\bibitem{eck2}   Ecksten, J., Svaiter, B.F., \emph{General projective splitting methods for sum of maximal monotone operators},  SIAM J. Control Optim., \textbf{48} (2009), 787-811.
		
		\bibitem{FicG}   Fichera, G., \emph{Sul problema elastostatico di signorini con ambigue condizioni al contorno},  Atti Accad. Naz. Lincei Rend. Cl. Sci. Fis. Mat. Nat., \textbf{34} (8) (1963), 138-142.
		
\bibitem{tse1}  Gibali, A.,  Thong, D.V., \emph{Tseng type methods for solving inclusion problems and its applications},  Calcolo \textbf{55}(2018),  1–22.
\bibitem{Godwin}   Godwin, E.C.,   Alakoya, T.O.,  Mewomo, O.T.,  Yao,  J.-C., \emph{Relaxed inertial Tseng extragradient method for variational inequality and fixed point problems}, {\it Appl. Anal.}, (2022), DOI:10.1080/00036811.2022.2107913.
\bibitem{Godwin1}  Godwin, E.C., Izuchukwu, C.,  Mewomo, O.T., \emph{Image restoration using a modified relaxed inertial method for  generalized split feasibility problems,}
 Math. Methods Appl. Sci., (2022), DOI:10.1002/mma.8849.
			
\bibitem{kaz}  Kazmi, K.R., Rizvi, S.H., \emph{Iterative approximation of a common solution of a split generalized equilibrium problem and a fixed point problem for nonexpansive semigroup},  Math. Sci., \textbf{7}, 1 (2013), Art. 1, 10 pp.
		
	\bibitem{KiXu}   Kim, T.H.,  Xu, H.H., \emph{Strong convergence of modified Mann iterations for asymptotically
		nonexpansive mappings and semigroups}, Nonlinear Anal., \textbf{64},  (2006), 1140-1152.
		
		
		\bibitem{LoKaF} Lohawech, P., Kaewcharoen, A.,  Farajzadeh, A., \emph{Algorithms for the common solution of the split variational inequality problems and ﬁxed point problems with applications}, J. Inequal. Appl., \textbf{2018} (358), (2018).	
		
		\bibitem{lor} Lorenz, D.A., Pock, T., \emph{An inertial forward-backward algorithm for monotone inclusions}, {\it J. Math. Imaging Vis.},\textbf{51} (2015), 311-325.
		
		\bibitem{mart1}   Martinet, B., \emph{R\'egularisation, din\'equations variationelles par approximations succesives}, 
Rev. Francaise Informat., Recherche Operationelle 4, Ser. R-3, pp.154-159.
		
		\bibitem{mou} Moudafi,A., Thera, M., \emph{Finding a zero of the sum of two maximal monotone operators},  J. Optim. Theory Appl., \textbf{94} (2) (1997), 425-448.
		
		\bibitem{mou1}  Moudafi, A., Oliny, M., \emph{Convergence of a splitting inertial proximal method for monotone operators},  J. Comput. Appl. Math., \textbf{155} (2003), 447-454.
		
		\bibitem{Nak} Nakajo, K., Takahashi, W., \emph{Strong convergence theorems for nonexpansive mappings and nonexpansive semigroups},
 J. Math. Anal. Appl.,\textbf{279} (2003), 372-379.
		

\bibitem{ogwo} Ogwo, G.N., Alakoya, T.O.,  Mewomo, O.T., \emph{Inertial Iterative Method With Self-Adaptive Step Size for Finite Family of Split Monotone Variational Inclusion and Fixed Point Problems in Banach Spaces},   Demonstr. Math., {\bf 55} (1), (2022), 193-216 .
\bibitem{ogwo1}  Ogwo, G.N.,  Alakoya, T.O., Mewomo, O.T., \emph{Iterative algorithm with self-adaptive step size for approximating the common solution of variational inequality and fixed point problems},  {\it Optimization}, (2021). DOI:10.1080/02331934.2021.1981897.

\bibitem{ogwo2}  Ogwo, G.N., Alakoya, T.O.,  Mewomo, O.T. \emph{ An inertial subgradient extragradient method with Armijo type step size for pseudomonotone variational inequalities with non-Lipschitz operators in Banach spaces},  J. Ind. Manag. Optim., (2022), doi:10.3934/jimo.2022239.
\bibitem{ogwo3}  Ogwo, G.N., Izuchukwu, C.,  Mewomo, O.T., \emph{Relaxed inertial methods for solving split variational inequality problems without product space formulation},
  Acta Math. Sci. Ser. B (Engl. Ed.), {\bf 42} (5) (2022), 1701–1733.

\bibitem{olona}   Olona, M.A.,  Alakoya, T.O.,  Owolabi,  A. O.-E.,  Mewomo,O.T.,  \emph{Inertial shrinking projection algorithm with self-adaptive step size for split generalized equilibrium and fixed point problems for a countable family of nonexpansive multivalued mappings},  Demonstr. Math., \textbf{54} (2021), 47-67.
		
		\bibitem{ma}  Ma, Z., Wang, L., Chang, S.S., Duan, W., \emph{Convergence theorems for split equality mixed equilibrium problems with applications},  Fixed Point Theory Appl., \textbf{2015} (2015), Art. 31
		
		\bibitem{marn}  Marino, G.,  Xu, H.H., \emph{Weak and strong convergence theorems for pseudo-contraction in Hilbert spaces},  J. Math. Anal. Appl., \textbf{329} (2007), 336-346.
\bibitem{owo}  Owolabi,  A. O.-E.,  Alakoya, T.O.,  Taiwo, A., Mewomo, O.T., \emph{A new inertial-projection algorithm for approximating common solution of variational inequality and fixed point problems of multivalued mappings},  Numer. Algebra Control Optim., {\bf 12} (2) (2022), 255–278.		
		\bibitem{phu} Phuengrattana, W., Lerkchaiyaphum, K., \emph{On solving the split generalized equilibrium problem and the fixed point problem for a countable family of nonexpansive multivalued mappings},  Fixed Point Theory Appl., \textbf{2018} (2018), Art. 6
		
		\bibitem{rock}   Rockafellar, R.T., \emph{Monotone operators and the proximal point algorithm},  SIAM J. Control Optim., \textbf{14} (1976), 877-898.
		
		\bibitem{ShZe}   Shahazad, N., Zegeye, H., \emph{Approximating of common point of fixed points of a pseudo-contractive mapping and zeros of sum of monotone mappings},  Fixed Point Theory Appl.,  \textbf{2014} (2014), Art. 85.
		
		\bibitem{sitt}   Sitthithakerngkiet, K.,  Deepho, J., Martínez-Moreno, J., Kumam, P., \emph{An iterative approximation scheme for solving a split generalized equilibrium, variational inequalities and fixed point problems}, Int. J. Comput. Math., \textbf{94} (12)(2017), 2373 – 2395.
		
		\bibitem{StampG}   Stampacchia, G., \emph{Formes bilinearies coercitives sur les ensembles convexes}, Acad. Sci. Paris, \textbf{258} (1964), 4413-4416.
		
		\bibitem{suant}  Suantai, S.,  \emph{Weak and strong convergence criteria of Noor iterations for asymptotically nonexpansive mappings},  J. Math. Anal. Appl., \textbf{311} (2005), 506 – 517.
\bibitem{taiwo}  Taiwo, A.,  Owolabi,A. O.-E.,  Jolaoso, L.O., Mewomo, O.T., Gibali, A.,  \emph{A new approximation scheme for solving various split inverse problems},
 Afr. Mat., {\bf 32} (3-4) (2021), 369–401.
		
\bibitem{tho}  Thong, D.V., Cholamjiak, P., \emph{Strong convergence of a forward-backward splitting method with a new step size for solving monotone inclusions},  Comput. Appl. Math., {\bf 38} (2) (2019), Paper No. 94, 16 pp.
		
		\bibitem{tse}  Tseng, P., \emph{A modified forward-backward splitting method for maximal method for maximal monotone mappings},
	 SIAM J. Control Optim., \textbf{38} (2000), 431-446.
\bibitem{yuy}  Yuying, T.,  Plubtieng, S.,  \emph{Strong convergence theorems by hybrid and shrinking projection methods for sums of two monotone operators},  J. Ineq. Appl., \textbf{2017} (2017), Art. 72.
\bibitem{uzor} Uzor, V.A.,  Alakoya, T.O., Mewomo, O.T., \emph{Strong convergence of a self-adaptive inertial Tseng's extragradient method for pseudomonotone variational inequalities and fixed point problems}, Open Math., {\bf 20}, (2022), 234–257.
 \bibitem{uzor1}  Uzor, V.A.,  Alakoya, T.O., Mewomo, O.T., \emph{On Split Monotone Variational Inclusion Problem with Multiple Output Sets with fixed point constraints},  Comput. Methods Appl. Math., (2022), DOI: 10.1515/cmam-2022-0199.

\bibitem{Zho}  Zhou, Y., \emph{Convergence theorems of fixed points for $k$-strict pseudo-contractions in Hilbert spaces},  Nonlinear Anal., \textbf{69} (2008), 456-462.

\end{thebibliography}

%
\setcounter{page}{1}
\setcounter{firstpage}{1}
\setcounter{lastpage}{4}
\renewcommand{\currentvolume}{??}
\renewcommand{\currentyear}{??}
\renewcommand{\currentissue}{??}
%
\title[A modified inertial shrinking projection algorithm]{A modified inertial shrinking projection algorithm with adaptive step size for solving split generalized equilibrium, monotone inclusion and fixed point problems}
\author{A- O. E. Owolabi}
\address{School of Mathematics, Statistics and Computer Science, \\  University of KwaZulu-Natal,\\
Durban, South Africa}
\email{218086824@stu.ukzn.ac.za}
%
\author{T. O. Alakoya}
\address{School of Mathematics, Statistics and Computer Science, \\  University of KwaZulu-Natal,\\
Durban, South Africa}
\email{alakoyat1@ukzn.ac.za, timimaths@gmail.com}

%
\author{O. T. Mewomo}
\address{School of Mathematics, Statistics and Computer Science, \\  University of KwaZulu-Natal,\\
Durban, South Africa}
\email{mewomoo@ukzn.ac.za}
%
\subjclass{65K15, 47J25, 65J15}
\keywords{Split generalized equilibrium problem, Monotone inclusion problem, Inertial method, Fixed point problem, Strict Pseudo-contractions, Multivalued mappings.}
\begin{abstract}
In this paper, we study the common solution problem of split generalized equilibrium problem, monotone inclusion problem and common fixed point problem for a countable family of strict pseudo-contractive multivalued mappings. We propose a modified shrinking projection algorithm of inertial form with self-adaptive step sizes for finding a common solution of the aforementioned problem. The self-adaptive step size eliminates the difficulty of computing the operator norm while the  inertial term accelerates the rate of convergence of the  proposed algorithm. Moreover, unlike several of the existing results in the literature, the monotone inclusion problem considered is a more general problem involving the sum of Lipschitz continuous monotone operators and maximal monotone operators, and knowledge of the Lipschitz constant is not required to implement our algorithm. Under some mild conditions, we establish strong convergence result for the proposed method. Finally, we present some applications and numerical experiments to illustrate the usefulness and applicability of our algorithm as well as comparing it with some related methods. Our results improve and extend  corresponding results in the literature.

\end{abstract}
\maketitle

\section{Introduction}
	\noindent Let $H$ be a real Hilbert space with induced norm $\|\cdot\|$ and inner product $\langle \cdot, \cdot \rangle$. Let $C$ be a nonempty closed convex subset of a real Hilbert space and let $F : C \times C \rightarrow \mathbb{R}$  be A bifunction. The \textit{equilibrium problem} (shortly, (EP)) in the sense of Blum and Oettli \cite{BlOe} is to find $\hat{x}\in C$ such that
	\begin{equation}\label{c1x}
		F(\hat{x},y)\geq 0,\quad \forall~ y\in C.
	\end{equation}
	\noindent The set of all solutions of EP \eqref{c1x} is denoted by $EP(F).$ The EP attracts considerable research efforts and serves as a unifying framework for studying many well-known
	problems, such as the Nonlinear Complementarity Problems (NCPs), Optimization Problems (OPs), Variational Inequality Problems (VIPs), Saddle Point Problems (SPPs), the Fixed Point Problem (FPP), the Nash equilibria and many others, and has many applications in physics and economics, (see, for example \cite{alakoya,CeYa,CeGiRe,ogwo,ogwo1,uzor} and the references therein).
	
	\noindent On the other hand, the {\it generalized equilibrium problem} (GEP) is defined as finding a point $x \in C$ such that
	\begin{equation}\label{c2x}
		F(x, y) + \phi(x, y) \geq 0, \forall y \in C,
	\end{equation} 	
	
	
	\noindent where $F, \phi : C \times C \rightarrow \mathbb{R}$ are bifunctions. We denote  the solution set of GEP \eqref{c2x} by $GE{P(F, \phi)}$. If $\phi = 0$, then the GEP \eqref{c2x} reduces to the equilibrium problem \eqref{c1x}.
	
	\noindent Let $C \subseteq H_1$ and $Q \subseteq H_2$ where $H_1$ and $H_2$ are real Hilbert spaces. Let $F_1, \phi_1 : C \times C \rightarrow \mathbb{R}$ and
	$F_2, \phi_2 : Q \times Q \rightarrow \mathbb{R}$ be nonlinear bifunctions, and $A : H_1 \rightarrow H_2$ be a bounded linear operator. The {\it split generalized equilibrium problem} (SGEP) introduced by Kazmi and Rizvi \cite{kaz} is defined as follows: Find $\bar{x} \in C$ such that
	\begin{equation}\label{c3x}
		F_1(\bar{x}, x) + \phi_1(\bar{x}, x) \geq 0, \forall x \in C,
	\end{equation}
	and such that
	\begin{equation}\label{c4x}
		\bar{y} = A\bar{x} \in Q~~\text{solves}~~F_2(\bar{y}, y) + \phi_2(\bar{y}, y) \geq 0, ~~\forall y \in Q.
	\end{equation}	
	\noindent The solution set of the split generalized equilibrium problem is denoted by
	\begin{equation}\label{c*}
		SGEP(F_1, \phi_1, F_2, \phi_2) = \{\bar{x} \in C : \bar{x} \in GEP(F_1, \phi_1)~~\text{and}~~A\bar{x} \in GEP(F_2, \phi_2)\}.
	\end{equation}	
	\noindent If $\phi_1 = 0$ and $\phi_2 = 0$, we obtain a special case of the split generalized equilibrium problem \eqref{c3x}-\eqref{c4x} called the {\it split equilibrium problem} (SEP) which is defined as follows:
	\begin{equation}\label{c5x}
		F_1(\bar{x}, x) \geq 0, \forall x \in C,
	\end{equation}
	and such that
	\begin{equation}\label{c6x}
		\bar{y} = A\bar{x} \in Q~~\text{solves}~~F_2(\bar{y}, y)  \geq 0, ~~\forall y \in Q.
	\end{equation}	 	
	\noindent We denote the solution set of the SEP \eqref{c5x}-\eqref{c6x}  by $\Omega := \{\bar{x} \in EP(F_1) : A\bar{x} \in EP(F_2)\}$. The split generalized equilibrium problem has been studied by numerous authors and several iterative algorithms have been proposed by many authors for solving the problem (see, \cite{phu, sitt}).
	
	\noindent Another important problem that we consider is the \textit{monotone inclusion problem} (MIP), which is defined as finding a point $z\in H$ such that
	\begin{equation}\label{abx}
		0 \in (B + D)z,
	\end{equation}
	where $B : H\to H$ is a nonlinear operator and $D : H\to 2^H$ is a set-valued operator. We denote the set of solutions of \eqref{abx} by  $(B+D)^{-1}(0).$ The MIP \eqref{abx} and related optimization problems have been studied by several authors with various iterative algorithms proposed for approximating their solutions in Hilbert spaces and  Banach spaces (see, for instance \cite{alakoya2,mou, Nak,tse, yuy,uzor1}). One of the most efficient methods for solving the MIP is the forward-backward splitting method (see \cite{atto,bot, cho, eck1, eck2, lor}).
	
	\noindent Martinez \cite{mart1} first introduced the Proximal Point Algorithm (PPA) for finding the zero point of a maximal monotone operator $B$. The sequence generated by  PPA is defined as follows:
	\begin{equation*}
		x_{n+1} = J_{r_n}^Dx_n,
	\end{equation*}
	where $0 < r_n < \infty$, $J_{r_n}^D = (I + r_nD)^{-1}$ is the resolvent operator of $D$ and $I$ is the identity mapping. This algorithm was eventually modified by Rockafellar \cite{rock} to the following PPA with errors:
	\begin{equation*}
		x_{n+1} = J_{r_n}^Dx_n + e_n,
	\end{equation*}
	where $\{e_n\}$ is an error sequence. It was proved that if $e_n\to 0$ such that $\sum\limits_{n=1}^{\infty}\|e_n\| < +\infty$, and the solution set $D^{-1}(0) \neq \emptyset$ and $\liminf\limits_{n\to\infty}r_n > 0$, then the sequence $\{x_n\}$ converges weakly to a zero point of $D.$
	
	\noindent Also, Moudafi and Th\'era \cite{mou} introduced the following iterative algorithm for solving MIP \eqref{abx}:
	\begin{equation}
		\begin{cases}
			x_n = J^D_{r}v_n,\\
			v_{n+1} = tv_n + (1 - t)x_n - \mu(1 - t)Bx_n,
		\end{cases}
	\end{equation}
	where $t \in (0,1),$ $r>0$, $B$ is Lipschitz continuous and strongly monotone and $D$ is maximal monotone. They proved that the sequence $\{x_n\}$ generated by the iterative algorithm converges weakly to an element in $(B + D)^{-1}(0)$.
	
	\noindent Alvarez and Attouch \cite{alv} proposed the following  modified PPA of inertial form:
	\begin{equation}\label{eq1.8}
		\begin{cases}
			y_n = x_n + \mu_n(x_n - x_{n-1}),\\
			x_{n+1} = J_{\lambda_n}^Dy_n, \qquad n\geq 1,
		\end{cases}
	\end{equation}
	where $\{\mu_n\}\subset [0,1)$, $\{\lambda_n\}$ is non-decreasing and
	\begin{equation}\label{eq1.9}
		\sum_{n=1}^{\infty}\mu_n\|x_n - x_{n-1}\|^2 < \infty,\quad \forall \mu_n < \frac{1}{3}.
	\end{equation}
	It was proved that Algorithm \eqref{eq1.8} converges weakly to a zero of $D$.
	
	\noindent Recently,  Moudafi and Oliny \cite{mou1} introduced the following inertial PPA for approximating the zero point problem of the sum of two monotone operators:
	\begin{equation}\label{eq1.10}
		\begin{cases}
			y_n = x_n + \mu_n(x_n - x_{n-1}),\\
			x_{n+1} = J_{\lambda_n}^D(y_n - \lambda_nBx_n), n\geq 1,
		\end{cases}
	\end{equation}
	where $D : H\to 2^H$ is maximal monotone and $B$ is Lipschitz continuous. They proved that the sequence generated by  Algorithm \eqref{eq1.10} converges weakly if $\lambda_n < \frac{2}{L}$, where $L$ is the Lipschitz constant of $B$.
	
	\noindent Moreover, the following inertial forward-backward algorithm was introduced by Lorenz and Pock \cite{lor}:
	\begin{equation}\label{eq1.11}
		\begin{cases}
			y_n = x_n + \mu_n(x_n - x_{n-1}),\\
			x_{n+1} = J_{\lambda_n}^D(y_n - \lambda_nBy_n), n\geq 1,
		\end{cases}
	\end{equation}
	where $\{\lambda_n\}$ is a positive real sequence. Algorithm \eqref{eq1.11} differs from Algorithm \eqref{eq1.10} since the operator $B$ is evaluated as the inertial extrapolate $y_n.$ The proposed algorithm was also proved to converge weakly to a solution of the MIP \eqref{abx}.
	

	\noindent In 2016, Deepho \cite{dph} introduced the general Ces\'aro mean iterative method for approximating a common solution of split generalized equilibrium, fixed point of nonexpansive mappings $T_j$ and variational inequality problems:
	
	\begin{equation}\label{algdep}
		\begin{cases}
			z_n = T_{r_n}^{(F_1, \phi_1)}(x_n + \gamma A^*(T_{r_n}^{(F_2, \phi_2)} - I)Ax_n),\\
			u_n = P_C(z_n - \lambda_nGz_n),\\
			x_{n+1} = \alpha_n\eta f(x_n) + \beta x_n + ((1 - \beta_n)I - \alpha_nK)\frac{1}{n+1}\sum_{j = 0}^{n} T_ju_n, ~~\forall n \geq 0,
		\end{cases}
	\end{equation}
	where $\{\alpha_n\}, \{\beta_n\} \subset (0, 1)$, $\{\lambda_n\} \in [a, b] \subset (0, 2\beta)$ and $\{r_n\} \subset (0, \alpha)$ and $\gamma \in \bigl(0, \frac{1}{L}\bigr)$, $L$ is the spectral radius of the operator $A^*A$ and $A^*$ is the adjoint of $A$. Under the following conditions:
	
	\begin{enumerate}
		\item[(C1)] $\lim_{n \to \infty}\alpha_n = 0,$ $\sum_{n = 0}^{\infty}\alpha_n = \infty;$\\
		\item[(C2)] $ 0 \leq \liminf_{n\to \infty}\beta_n \leq \limsup_{n \to \infty}\beta_n < 1;$\\
		\item[(C3)] $\lim_{n \to \infty}|\lambda_{n + 1} - \lambda_n| = 0;$\\
		\item[(C4)] $\liminf_{n\to \infty}r_n > 0, \lim_{n \to \infty}|r_{n+1} - r_n| = 0.$
	\end{enumerate}
	\noindent the authors proved that the sequence $\{x_n\}$ converges strongly to an element $q$ in the solution set $\Omega,$ where $q = P_{\Omega}(I - K + \gamma f)(q)$ is the unique solution of the variational inequality problem
	
	\begin{equation*}
		\langle (K - \gamma f)q, x - q \rangle \geq 0,~~\forall x \in \Omega.
	\end{equation*}
	
	\noindent Also, in 2017, Sitthithakerngkiet \cite{sitt} proposed and studied the following iterative method for approximating a common solution of split generalized equilibrium, variational inequality for an inverse-strongly monotone mapping and fixed point problems of nonexpansive mappings in Hilbert spaces:
	
	\begin{equation}\label{algsitth}
		\begin{cases}
			z_n = T_{r_n}^{(F_1, \phi_1)}(x_n + \gamma A^*(T_{r_n}^{(F_2, \phi_2)} - I)Ax_n),\\
			x_{n+1} = \alpha_nf(x_n) + \beta_nx_n + \xi_nT[\sigma_nv + (1 - \sigma_n)P_C(z_n - \lambda_nGz_n)],
		\end{cases}
	\end{equation}
	where $v\in C$ is a fixed point, $r_n \in (0, \infty), \mu \in \bigl(0, \frac{1}{L}\bigr)$, $L$ is the spectral radius of the operator $A^*A$, $A^*$ is the adjoint of $A$, sequences $\{\alpha_n\}, \{\beta_n\}, \{\xi_n\}$ and $\{\sigma_n\}$ are in $(0, 1)$ and satisfy $\alpha_n + \beta_n + \xi_n = 1, \lambda_n \in [a, b]$ for some $a, b$ with $0 < a < b < 2\beta_n$ and $\{\gamma_n\} \subset [c, 1]$ for some $c \in (0, 1)$. Assume that the following conditions are satisfied:
	
	\begin{enumerate}
		\item[(C1)] $\lim_{n \to \infty}\alpha_n = 0,$ $\sum_{n = 0}^{\infty}\alpha_n = \infty;$\\
		\item[(C2)] $\lim_{n\to \infty}\sigma_n = 0;$\\
		\item[(C3)] $ 0 \leq \liminf_{n\to \infty}\beta_n \leq \limsup_{n \to \infty}\beta_n < 1;$\\
		\item[(C4)] $\lim_{n \to \infty}|\lambda_{n + 1} - \lambda_n| = 0;$\\
		\item[(C5)] $\liminf_{n\to \infty}r_n > 0, \lim_{n \to \infty}|r_{n+1} - r_n| = 0,$
	\end{enumerate}
	\noindent the authors proved that  the sequence $\{x_n\}$ converges strongly to $z \in \Omega,$ where $z = P_{\Omega}f(z).$
	
	
	
	
	\noindent Recently, Phuengrattana and Lerkchaiyaphum \cite{phu} introduced the following shrinking projection method for solving SGEP and FPP for a countable family of nonexpansive  multivalued mappings: For $x_1\in C$ and $C_1=C,$ then
	\begin{equation} \label{algphu}
		\begin{cases}
			z_n= T^{(F_1, \phi_1)}_{r_n} (I - \gamma A^*(I - T^{(F_2, \phi_2)}_{r_n}) A)x_n, \\
			y_n = \delta_{n, 0}x_n + \sum_{j = 1}^n\delta_{n, j}u_{n, j},\quad\quad u_{n,j} \in P_j z_n,\\
			C_{n+1} = \{ p \in C_n : \|y_n - p \|^2 \le \| x_n - p\|^2 \},\\
			x_{n+1} = P_{C_{n+1}} x_1,\quad\quad n\in \mathbb{N}.
		\end{cases}
	\end{equation}
	They proved that if
	\begin{enumerate}
		\item[(i)] $\lim \inf _{n \rightarrow \infty}$ $r_n > 0$,
		\item[(ii)] The limits $\lim_{n\rightarrow \infty}\delta_{n,j} \in (0,1)$ exist for all $j \ge 0$,
	\end{enumerate}
	then the sequence $\{x_n\}$ generated by $\eqref{algphu}$ converges strongly to $P_{\Gamma}x_1$, where $\Gamma = \bigcap_{j=1}^ \infty$ $F(P_j)\cap SGEP(F_1,\phi_1, F_2, \phi_2) \ne \emptyset,$ $F(P_j)$ is the set of fixed points of $P_j$ and $P_j$ is a countable family of nonexpansive multivalued mappings.
	
	\noindent In 2021, Olona et al. \cite{olona} proposed an inertial shrinking projection defined as follows for split generalized equilibrium and fixed point problems for a countable family of nonexpansive multivalued mappings : for $x_0, x_1 \in C$ with $C_1 = C$, then
	
	\begin{equation} \label{xii}
		\begin{cases}
			w_n = x_n + \theta_n(x_n - x_{n-1}),\\
			u_n = T^{(F_1, \phi_1)}_{r_n}(I - \gamma_n A^*(I - T^{(F_2, \phi_2)}_{r_n})A)w_n,\\
			z_n = \delta_{n,0}u_n + \sum_{i=1}^{n} \delta_{n,j}y_{n,j},~~ y_{n,j} \in P_ju_n,\\
			C_{n+1} =\{ p \in C_n : \|z_n - p\|^2 \le \|x_n - p\|^2 \\
- 2\theta_n\langle x_n - p, x_{n-1}-x_n \rangle + \theta_n^2\|x_{n-1} - x_n\|^2\},\\
			x_{n+1} = P_{C_{n+1}} x_1,\quad n \in \mathbb{N},
		\end{cases}
	\end{equation}
	
	\begin{equation*}
		\gamma_n =
		\begin{cases}
			\frac{\tau_n \|(I- T_{r_n}^{(F_2, \phi_2)}) Aw_n \|^2}{\| A^* (I - T^{(F_2, \phi_2)}_{r_n}) Aw_n\|^2}\quad\quad \text{if}~~~ Aw_n \ne\	 T_{r_n}^{(F_2, \phi_2)} Aw_n,\\
			\gamma\hspace{100pt}	\text{otherwise ($\gamma$ being any nonnegative real number)},
		\end{cases}
	\end{equation*}
	\noindent where $A : H_1 \to H_2$ is a bounded linear operator, $0<a\leq\tau_n\leq b<1, \{\theta_n\} \subset \mathbb{R},$ $\{\delta_{n,j}\} \subset (0,1),$ such that $\sum_{j=0}^{n}\delta_{n,j} = 1,$ and $\{r_n\} \subset (0,\infty).$ $\{P_j\}$ is a countable family of nonexpansive multivalued  mappings,  $F_1,\phi_1:C\times C\rightarrow\mathbb{R},~ F_2,\phi_2:Q\times Q\rightarrow\mathbb{R}$ are bifunctions. Under some appropriate conditions, it was proved that the sequence $\{x_n\}$ converges strongly to $P_\Omega x_1,$ where $\Omega = \bigcap_{j=1}^{\infty} F(P_j) \cap SGEP (F_1,\phi_1, F_2, \phi_2) \neq \emptyset$.
	
	
	\noindent Motivated by the above results and the current research interest in this direction, in this paper, we propose a new iterative algorithm of inertial type with self-adaptive step size for approximating the common solution of SGEP \eqref{c3x}-\eqref{c4x}, MIP \eqref{abx} and FPP of strictly pseudo-contractive multivalued mappings. We prove that the sequence generated by our algorithm converges strongly to a solution of the investigated problem. Finally, we present some applications and numerical examples to illustrate the usefulness and efficiency of the proposed method in comparison with some related methods. Our proposed method uses self-adaptive step size and employs inertial technique to accelerate the rate of convergence of the proposed method. The implementation of our proposed algorithm does not require a prior knowledge of the norm of the bounded linear operator.
	
	\noindent  Subsequent sections of this paper are organised as follows: In Section \ref{Sec2}, we recall some basic definitions and lemmas that are relevant in establishing our main results. In Section \ref{Sec3}, we present our proposed algorithm and highlight some of its features. In Section \ref{Sec4}, we prove some lemmas that are useful in establishing the strong convergence of our proposed algorithm and also prove the strong convergence theorem for the algorithm. In Section \ref{Sec5}, we apply our result to study some optimization problems while in Section \ref{Sec6}, we present some numerical experiments to illustrate the performance of our method and compare it with some related methods in the literature. Finally, in Section \ref{Sec7} we give a concluding remark.
	
	
	\section{Preliminaries}\label{Sec2}
	\noindent Let $C$ be a nonempty, closed and convex subset of a real Hilbert space $H$ with inner product $\langle \cdot, \cdot \rangle$ and norm $\|\cdot\|$. We denote $x_n \rightarrow x$ to mean that sequence $\{x_n\}$ converges strongly to $x$ and $x_n \rightharpoonup x$ to indicate that the sequence $\{x_n\}$ converges weakly to $x.$ We write $w_\omega(x_n)$ to denote set of weak limits of $\{x_n\},$ that is,
	$$\omega_w(x_n):= \{x\in H: x_{n_j}\rightharpoonup x~ \text{for some subsequence}~ \{x_{n_j}\}~ \text{of} ~\{x_{n}\}\}.$$ The nearest point projection of $H$ onto $C$ denoted by $P_C$ is defined for each $x\in H,$ as the unique element $P_Cx\in C$ such that
	\begin{equation}
		\|x - P_Cx\| \leq \|x - y\|,~~\forall~y \in C.
	\end{equation}
	\noindent It is well known that $P_C$ is nonexpansive and has the following  characteristics (see \cite{alakoya1,Godwin}:
	
	\begin{equation}
		\|P_Cx - P_Cy\|^2 \leq \langle x - y, P_Cx - P_Cy \rangle,~~\forall~x, y \in H_1,
	\end{equation}
	
	\begin{equation}
		\langle x - P_Cx, y - P_Cx \rangle \leq 0,	
	\end{equation}
	
	\begin{equation}
		\|x - y\|^2 \leq \|x - P_Cx\|^2 + \|y - P_Cx\|^2,~~\forall x \in H, y \in C,
	\end{equation}
	
	\begin{equation}
		\|(x - y) - (P_Cx - P_Cy)\|^2 \geq \|x - y\|^2 - \|P_Cx - P_Cy\|^2,~~x, y \in H.	
	\end{equation}
	
	\noindent A mapping $B : C \rightarrow H$ is said to be monotone if
	\begin{equation}
		\langle Bu - Bv, u - v \rangle \geq 0,~~\forall u, v \in C.
	\end{equation}
	\noindent Moreover, if $B$ satisfies
	\begin{equation}
		\langle Bu - Bv, u - v \rangle \geq \alpha\|Bu - Bv\|^2,~~\forall u, v \in C,
	\end{equation}
	for some positive real number $\alpha$. Then, $B$ is called an $\alpha$-inverse-strongly monotone  mapping. It is clear that every inverse-strongly monotone mapping is monotone.   	
	\begin{lemma}\cite{olona, marn}\label{lem3f}
		Let $H$ be a real Hilbert space, $\lambda \in \mathbb{R}$, then $\forall x, y \in H,$ we have
		\begin{enumerate}
			\item[(i)] $\|x + y\|^2 = \|x\|^2 + 2\langle x, y \rangle + \|y\|^2$;
			\item[(ii)] $\|x - y\|^2 = \|x\|^2 - 2\langle x, y \rangle + \|y\|^2$;
			\item[(iii)] $\|x + y\|^2 \leq \|x\|^2 + 2\langle y, x + y \rangle$;
			\item[(iv)] $\|\lambda x + (1 - \lambda)y\|^2 = \lambda\|x\|^2 + (1 - \lambda)\|y\|^2 - \lambda(1 - \lambda)\|x - y\|^2$.
		\end{enumerate}
	\end{lemma}
	
	\begin{lemma}\label{lem3a1}\cite{Nak}
		Let $C$ be a nonempty closed convex subset of a real Hilbert space $H$, and let
		$P_C : H \rightarrow C$ be the metric projection. Then
		
		\begin{equation*}
			\|y - P_Cx\|^2 + \|x - P_Cx\|^2 \leq \|x - y\|^2,~~\forall x \in H, y \in C.
		\end{equation*}
		
	\end{lemma}
	
	\begin{lemma}\label{lem3b}\cite{ZeSh}
		Let $x_i \in H, (1 \leq i \leq m), \sum_{i=1}^{m}\alpha_i = 1$, where $\{\alpha_i\} \subseteq (0, 1)$. Then
		\begin{equation*}
			\bigg\|\sum_{i=1}^{m}\alpha_ix_i\bigg\|^2 = \sum_{i=1}^{m}\alpha_i\|x_i\|^2 - \sum_{1\le i<j\le m}\alpha_i\alpha_j\|x_i - x_j\|^2.
		\end{equation*}
	\end{lemma}
	
	
	
	\begin{lemma}\cite{KiXu}\label{Lem2.4}
		Let $C$ be a nonempty, closed and convex subset of a real Hilbert space $H.$ Given $x,y,z\in H$ and $a\in\mathbb(R),$ the set $D = \{v\in C: \|y-v\|^2\le \|x-v\|^2 + \langle z,v \rangle + a \}$ is convex and closed.
		
		
	\end{lemma}
	
	
	
	
	\begin{assumption}\label{Ass 1}
		Let $C$ be a nonempty closed convex subset of a Hilbert space $H.$ Let $F_1 : C \times C \rightarrow \mathbb{R}$ and $\phi_1 : C \times C \rightarrow \mathbb{R}$ be two bifunctions that satisfy the following conditions:
		\begin{enumerate}
			\item [(A1)] $F_1(x,x) = 0$ for all $x \in C,$
			\item [(A2)] $F$ is monotone, that is, $F_1(x,y) + F_1(y,x) \le 0$ for all $x,y \in C,$
			\item[(A3)] $F$  is upper hemicontinuous, that is, for all $ x,y,z \in C$,\\
 $\lim_{t \downarrow 0} F\big(tz + (1-t)x,y \big)\le F(x,y),$
			\item [(A4)] for each $x \in C, y \mapsto F_1(x,y)$ is convex and lower semicontinuous,
			\item [(A5)] $\phi_1(x, x) \geq 0$, for all $x \in C,$
			\item[(A6)] for each $y \in C, x \mapsto \phi_1(x, y)$ is upper semicontinuous,
			\item[(A7)] for each $x \in C, y \mapsto \phi_1(x, y)$ is convex and lower semicontinuous,
		\end{enumerate}
		\noindent and assume that for fixed $r > 0$ and $z \in C$, there exists a nonempty compact convex subset $K$ of $H_1$ and $x \in C \cap K$ such that
		\begin{equation*}
			F_1(y, x) + \phi_1(y, x) + \frac{1}{r}\langle y - x, x - z \rangle < 0, ~~~\forall y \in C~~\backslash K.
		\end{equation*}
	\end{assumption}
	
	\begin{lemma}\label{lem3c}\cite{ma}
		Let $C$ be a nonempty closed convex subset of a Hilbert space $H$. Let $F : C \times C \rightarrow \mathbb{R}$ and $\phi_1 : C \times C \rightarrow \mathbb{R}$ be two bifunctions that satisfy Assumption \ref{Ass 1}. Assume that $\phi$ is monotone. For $r > 0$ and and $x \in H$. Define mapping $T_r^{(F, \phi)} : H \rightarrow C$ as follows:
		\begin{equation*}
			T_r^{(F, \phi)}(x) = \bigl\{z \in C: F(z, y) + \phi(z, y) + \frac{1}{r}\langle y - z, z - x \rangle \geq 0, ~~~\forall y \in C\bigr\}
		\end{equation*}
		\noindent for all $x \in H_1$. Then
		\begin{enumerate}
			\item[(1)] for each $x \in H_1$, $T_r^{(F, \phi)} \neq \emptyset,$
			\item[(2)] $T_r^{(F, \phi)}$ is single-valued,
			\item[(3)] $T_r^{(F, \phi)}$ is firmly nonexpansive, that is, for any $x, y \in H_1$,
			\begin{equation*}
				\|T_r^{(F, \phi)}x - T_r^{(F, \phi)}y\|^2 \leq \langle T_r^{(F, \phi)}x - T_r^{(F, \phi)}y, x - y\rangle,
			\end{equation*}
			\item[(4)] $F(T_r^{(F, \phi)}) = GEP(F, \phi),$
			\item[(5)] GEP$(F, \phi)$ is closed and convex.
			
		\end{enumerate}
		
	\end{lemma}
	
	
	\begin{lemma}\label{lem3g}\cite{suant}
		Let $X$ be a Banach space space satisfying Opial's condition and let $\{x_n\}$ be a sequence in $X$. Let $u, v \in X$ be such that
		\begin{equation*}
			\lim_{n\to \infty}\|x_n - u\|~~\text{and}~~\lim_{n\to \infty}\|x_n - v\|~~\text{exist}.
		\end{equation*}
		If $\{x_{n_k}\}$ and $\{x_{m_k}\}$ are subsequences of $\{x_n\}$ which converge weakly to $u$ and $v$, respectively, then $u = v$.
	\end{lemma}
	
	\begin{lemma}\label{lem3h}\cite{brez}
		Let $B : H \to 2^H$ be a maximal monotone mapping and $A : H \to H$ be a Lipschitz continuous and monotone mapping. Then, the mapping $A + B$ is a maximal monotone mapping.
	\end{lemma}
	
	\begin{lemma}\cite{tse1}
		Let $B : H \to 2^H$ be a maximal monotone operator and $A : H \to H$ be a mapping on $H$. Define $T_{\lambda} := (I + \lambda B)^{-1}(I - \lambda A)$, $\lambda > 0$. Then, we have the following
		\begin{equation}
			Fix(T_{\lambda}) = (A + B)^{-1}(0),~~\forall \lambda > 0.
		\end{equation}
		
	\end{lemma}
	
	
	
	\noindent Let $D$ be a nonempty subset of $H$. $D$ is said to be proximal if there exists  $y \in D$ such that
	\begin{equation*}
		\|x - y\| = d(x, D),~~x \in H.
	\end{equation*}
	\noindent Let $CC(C), CB(C)$ and $P(C)$ be the family of nonempty closed convex subset of $H$, nonempty closed bounded subsets of $H$ and nonempty proximal bounded subsets of $H$ respectively. The Hausdorff metric on $CB(C)$ is defined as follows:
	\begin{equation*}
		H(A, B) : = \max\biggl\{\sup_{x\in A} d(x, B), \sup_{y\in B} d(y, A) ,~~~\forall A, B \in CB(C)\biggr\}.
	\end{equation*}
	Let $S:C\rightarrow 2^C$ be a multivalued mapping.
	An element $x \in H$ is said to be a fixed point of $S$ if $x \in Sx.$ We say that $S$ satisfies the endpoint condition if $Sp=\{p\}$ for all $p\in F(S).$ For multivalued mappings $S_i: H\rightarrow 2^{H}~(i\in\mathbb{N})$ with $\cap_{i=1}^\infty F(S_i)\neq\emptyset,$ we say $S_i$ satisfies the common endpoint condition if $S_i(p)=\{p\}$ for all $i\in\mathbb{N},~ p\in\cap_{i=1}^\infty F(S_i).$
	
	
	\begin{definition}
		\noindent Let $A : H \to H$ be a nonlinear operator. Then $A$ is called
		
		\begin{enumerate}
		\item[(i)] Lipschitz continuous if for all $L > 0$
			\begin{equation*}
				\|Ax - Ay\| \leq L\|x - y\|,~~ \forall x, y \in H;
			\end{equation*}
			if $0 \leq L < 1$, then $A$ is a contraction mapping,	
		\item[(ii)] $\beta-$strongly monotone if for all $\beta > 0$
			\begin{equation*}
				\langle Ax - Ay, x - y \rangle \geq \beta\|x - y\|^2,~~ \forall x, y \in H.
			\end{equation*}
		\end{enumerate}
	\end{definition}
	
	\begin{definition}
		Let $S : C \to CB(C)$ be a multivalued mapping. $S$ is said to be
		
		\begin{enumerate}
			\item[(i)] nonexpansive if
			\begin{equation*}
				H(Sx, Sy) \leq \|x - y\|,~~ \forall x, y \in C,
			\end{equation*}
			\item[(ii)] quasi-nonexpansive if $F(S) \neq \emptyset$ such that
			\begin{equation*}
				H(Sx, Sp) \leq \|x - p\|,~~ \forall x \in C, p \in F(S),
			\end{equation*}
			\item[(iii)] $k$- strictly pseudo-contractive if there exists a constant $k \in [0, 1)$ such that
			\begin{equation}\label{st}
				(H(Sx, Sy))^2 \leq \|x - y\|^2 + k\|(x - u) - (y - v)\|^2,~~\forall u\in Sx, v \in Sy 	
			\end{equation}
			\noindent	If $k = 1$ in \eqref{st}, then the mapping $S$ is said to be pseudo-contractive.
		\end{enumerate}
		\noindent Clearly, the class of $k$-strict pseudo-contractive mappings properly contains the class of nonexpansive mappings. That is, $S$ is nonexpansive if and only if $S$ is $0$-strict pseudo-contractive. It is known that if $S$ is a $k$-strict pseudo-contraction and $F(S)\neq\emptyset,$ then $F(S)$ is a closed convex subset of $H$ (see \cite{Zho}). Strict pseudo-contractions have many applications, due to their ties with inverse strongly monotone operators. It is known that, if $B$ is a strongly monotone operator, then $S = I - B$ is a strict pseudo-contraction, and so we can recast a problem of zeros for $B$ as a fixed point problem for $S,$ and vice versa (see e.g. \cite{ChYa, ShZe}).
		
		\noindent Let $S : H \to CB(H)$ be a multivalued mapping. The multivalued mapping $I-S$ is said to be demiclosed at zero if for any sequence $\{x_n\}\subset H$ which converges weakly to $p$ and the sequence $\{\|x_n-u_n\|\}$ converges strongly to 0, where $u_n\in Sx_n,$ then $p\in F(S).$
	\end{definition}
	\section{Proposed Method}\label{Sec3}
	\noindent In this section, we present our proposed algorithm.
	
	\noindent Let $C$ and $Q$ be nonempty closed convex subsets of real Hilbert spaces $H_1$ and $H_2$, respectively. Let $A : H_1 \rightarrow H_2$ be a bounded linear operator, and let $\{S_i\}_{i=1}^m$ be a countable family of $k_i$-strictly pseudo-contractive multivalued mappings of $C$ into $CB(C)$ such that $I-S_i$ is demiclosed at zero for each $i=1,2,\ldots,m,$ $S_ip = \{p\}$ for each $p \in \cap_{i=1}^mF(S_i)$ and $k=\max\{k_i\}.$ Let $F_1, \phi_1 : C \times C \rightarrow \mathbb{R}, F_2, \phi_2 : Q \times Q \rightarrow $ be bifunctions satisfying Assumptions \ref{Ass 1}. Let $\phi_1, \phi_2$ be monotone, $\phi_1$ be upper hemicontinuous, and $F_2$ and $\phi_2$ be upper semicontinuous in the first argument. Let $B : H_1 \rightarrow H_1$ be $L$-Lipschitz continuous and monotone and $D : H_1 \rightarrow 2^{H_1}$ be a maximal monotone operator such that $\Gamma = SGEP(F_1, \phi_1, F_2, \phi_2)\cap  \cap_{i=1}^mF(S_i) \cap (B + D)^{-1}(0) \neq \emptyset.$ We establish the convergence of our algorithm under the following conditions on the control parameters:
	
	\begin{enumerate}
		\item[(C1)] $0 < a \leq \tau_n \leq b < 2, \{r_n\} \subset (0, \infty), \liminf_{n\to \infty}r_n > 0,$
		\item[(C2)] $\liminf_n\alpha_{n, i}(\alpha_{n, 0} - k)>0$ and $\lim_{n\to \infty}\alpha_{n, i} \in (0, 1)$ exists for all $i \geq 0$.
	\end{enumerate}
	Now, we present our proposed algorithm as follows:
	
	
	\hrule\hrule
	\begin{algorithm}\label{alg3a}\noindent
		\hrule
		
		\noindent \textbf{Initialization:} Select $x_0, x_1 \in H_1$, $s_1>0, \mu \in (0, 1),\theta_n \in [-\theta, \theta]$ for some $\theta>0$ and $C_1=C.$
		
		\noindent \textbf{Iterative Step:} Given the current iterate $x_n$, calculate the next iterate as follows:
		
		\noindent \textbf{Step 1 :} Compute
		\begin{equation*}
			w_n = x_n + \theta_n(x_n - x_{n - 1}).
		\end{equation*}
		\noindent \textbf{Step 2 :} Compute
		\begin{equation*}
			z_n = T_{r_n}^{(F_1, \phi_1)}(I - \gamma_nA^*(I - T_{r_n}^{(F_2, \phi_2)})A)w_n.
		\end{equation*}
		\noindent \textbf{Step 3 :} Compute
		\begin{equation*}
			y_n = \alpha_{n, 0}z_n + \sum_{i = 1}^m\alpha_{n, i}u_{n, i},\quad u_{n, i} \in S_iz_n.
		\end{equation*}
		\noindent \textbf{Step 4 :} Compute
		\begin{equation*}
			\begin{cases}
				v_n = (I + s_nD)^{-1}(I - s_nB)y_n=J_{s_n}^D(I - s_nB)y_n \\
				t_n = v_n - s_n(Bv_n - By_n)\\
				C_{n+1} = \{p \in C_n : \|t_n - p\|^2 \leq \|w_n - p\|^2 - \bigg(1 - \mu^2\frac{s_n^2}{s_{n+1}^2}\bigg)\|y_n - v_n\|^2\}\\
				x_{n+1} = P_{C_{n+1}}x_0,
			\end{cases}
		\end{equation*}
		\noindent \textbf{Step 5 :} Compute
		\begin{equation}\label{ck}
			s_{n+1} = \begin{cases}
				\min\biggl\{\frac{\mu\|y_n - v_n\|}{\|By_n - Bv_n\|} , s_n\biggr\} & \text{if}~~By_n - Bv_n \neq 0.\\
				s_n   & \text{otherwise},
			\end{cases}
		\end{equation}
		\noindent Set $n := n + 1$ and return to {\bf Step 1},\\
		
		\noindent where
		\begin{equation*}
			\gamma_n =
			\begin{cases}
				\tau_n\frac{||(I - T_{r_n}^{(F_2, \phi_2)})Aw_n||^2}{||A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n||^2} &  \text{If}~~ Aw_n \neq T^{(F_2, \phi_2)}_{r_n}Aw_n\\
				
				\gamma  &  \text{otherwise}~~(\gamma~~\text{being any non-negative real number}).
			\end{cases}
		\end{equation*}
		
	\end{algorithm}
	\hrule\hrule
	
	\begin{remark}
		\noindent We observe that
		\begin{enumerate}
		\item[(i)] The implementation of our proposed algorithm does not require prior knowledge of the operator norm.  Hence, this makes our method easily implementable.
		\item[(ii)] We employ the inertial technique to accelerate the rate of convergence.
		\item[(iii)] The underlying single-valued operator $B : H_1 \rightarrow H_1$ for most of the results on monotone inclusion problem in the literature are either strongly monotone or inverse strongly monotone while the single-valued operator in our proposed algorithm is only required to be monotone and Lipschitz continuous. Moreover, knowledge of the Lipschitz constant of the operator is
		not required to implement our proposed algorithm. Thus, our method is more applicable than several
		of the existing methods in the literature.
		
		\item[(iv)] Our result extends and improves on the results of Deepho et al. \cite{dph}, Sitthithakerngkiet et al. \cite{sitt}, Phuengrattana and Lerkchaiyaphum \cite{phu}, Olona et al. \cite{olona} and several other results in the current literature in this direction.
		\end{enumerate}
		
	\end{remark}
	
	
	\section{Convergence Analysis}\label{Sec4}
	\noindent In this section, we analyze the convergence of our proposed algorithm.	
	
	
	\begin{lemma}\label{Lem4.1}
		Let $\{s_n\}$ be a sequence generated by \eqref{ck}. Then,  $\{s_n\}$ is a nonincreasing sequence and
		
		\begin{equation}
			\lim_{n \to \infty}s_n = s \geq \min\biggl\{s_1, \frac{\mu}{L}\biggr\}.
		\end{equation}
	\end{lemma}
	\begin{proof}
		\noindent\\
		\noindent From \eqref{ck}, it is clear that $\{s_n\}$ is a nonincreasing sequence. Moreover, observe that if $By_n - Bv_n \neq 0,$ then
		
		\begin{equation}
			\frac{\mu\|y_n - v_n\|}{\|By_n - Bv_n\|} \geq \frac{\mu}{L}.
		\end{equation}
		Hence, the sequence $\{s_n\}$ has the lower bound $\min\biggl\{s_1, \frac{\mu}{L}\biggr\}$.
	\end{proof}
	
	\begin{lemma}\cite{tse1}\label{Lem4.2}
		Let $\{x_n\}$ be a sequence generated by Algorithm \ref{alg3a}. Then the following inequality holds for all $p\in\Gamma:$
	
		\begin{equation}\label{c1}
			\|t_n - p\|^2 \leq \|y_n - p\|^2 - \bigg(1 - \mu^2\frac{s_n^2}{s_{n+1}^2}\bigg)\|y_n - v_n\|^2, ~~p \in \Gamma,
		\end{equation}
		and
		\begin{equation}\label{c^2}
			\|t_n - v_n\| \leq \mu\frac{s_n}{s_{n+1}}\|y_n - v_n\|.
		\end{equation} 	

	\end{lemma}

	
\begin{proof}
		\noindent By the definition of $s_n$, we have
		\begin{equation}\label{csd1}
			\|By_n - Bv_n\| \leq \frac{\mu}{s_{n+1}}\|y_n - v_n\|\quad\forall~n\in\mathbb{N}.
		\end{equation}
		\noindent Clearly, if $By_n = Bv_n$, then \eqref{csd1} holds. Otherwise , we have
		
		\begin{equation*}
			s_{n+1} = \min\biggl\{\frac{\mu\|y_n - v_n\|}{\|By_n - Bv_n\|}, s_n\biggr\} \leq \frac{\mu\|y_n - v_n\|}{\|By_n - Bv_n\|}.
		\end{equation*}
		\noindent This implies that
		\begin{equation*}
			\|By_n - Bv_n\| \leq \frac{\mu}{s_{n+1}}\|y_n - v_n\|.
		\end{equation*}
		\noindent Thus, \eqref{csd1} holds when $By_n = Bv_n$ and $By_n\ne Bv_n.$ Let $p \in \Gamma,$ then by Lemma \ref{lem3f}, we have
		\begin{align}
			\|t_n - p\|^2 & = \|v_n - s_n(Bv_n - By_n) - p\|^2\nonumber\\
			& = \|v_n - p\|^2 + s^2_n\|Bv_n - By_n\|^2 - 2s_n\langle v_n - p, Bv_n - By_n \rangle\nonumber\\
			& = \|y_n - p\|^2 + \|y_n - v_n\|^2 + 2\langle v_n - y_n, y_n - p \rangle \nonumber\\
	        &+ s^2_n\|Bv_n - By_n\|^2 - 2s_n\langle v_n - p, Bv_n - By_n \rangle\nonumber\\
			& = \|y_n - p\|^2 + \|y_n - v_n\|^2 - 2\langle v_n - y_n, v_n - y_n \rangle + 2\langle v_n - y_n, v_n - p \rangle \nonumber\\
	       &+ s^2_n\|Bv_n - By_n\|^2 -2s_n\langle v_n - p, Bv_n - By_n\rangle\nonumber\\
			& = \|y_n - p\|^2 - \|y_n - v_n\|^2 + 2\langle v_n - y_n, y_n - p \rangle \nonumber\\
	       &+ s^2_n\|Bv_n - By_n\|^2 - 2s_n\langle v_n - p, Bv_n - By_n \rangle\nonumber\\
			& = \|y_n - p\|^2 - \|y_n - v_n\|^2 - 2\langle y_n - v_n - s_n(By_n - Bv_n), v_n - p\rangle \nonumber\\
	       &+ s^2_n\|Bv_n - By_n\|^2\label{cg}.
		\end{align}
		\noindent By applying \eqref{csd1} in \eqref{cg}, we obtain
		\begin{equation}
			\|t_n - p\|^2  \leq \|y_n - p\|^2 - \bigg(1 - \mu^2\frac{s^2_n}{s^2_{n+1}}\bigg)\|y_n - v_n\|^2 - 2\langle y_n - v_n - s_n(By_n - Bv_n), v_n - p \rangle \label{chl}.
		\end{equation}
		\noindent We now prove that $\langle y_n - v_n - s_n(By_n - Bv_n), v_n - p \rangle \geq 0.$
		\noindent Since $v_n = (I + s_nD)^{-1}(I - s_nB)y_n$, then we have $(I - s_nB)y_n \in (I + s_nD)v_n$. Recall that $D$ is maximal monotone. Then there exists $u_n \in Dy_n$ such that
		\begin{equation*}
			(I - s_nB)y_n = v_n + s_nu_n,
		\end{equation*}
		\noindent  from which we obtain
		\begin{equation}\label{dj}
			u_n = \frac{1}{s_n}(y_n - v_n - s_nBy_n).
		\end{equation}
		\noindent Moreover, we have $0 \in (B + D)p$ and $Bv_n + u_n \in (B + D)v_n.$ Since $B + D$ is maximal monotone, we get
		\begin{equation}\label{dxt}
			\langle Bv_n + u_n, v_n - p\rangle \geq 0.
		\end{equation}
		By substituting \eqref{dj} into \eqref{dxt}, we obtain
		\begin{equation*}
			\frac{1}{s_n}\langle y_n - v_n - s_nBy_n + s_nBv_n, v_n - p \rangle \geq 0.
		\end{equation*}
		This implies that
		\begin{equation}\label{dzf}
			\langle y_n - v_n - s_n(By_n- Bv_n), v_n - p \rangle \geq 0.
		\end{equation}
		\noindent By applying \eqref{dzf} in \eqref{chl}, we have
		\begin{equation}\label{lgi}
			\|t_n - p\|^2  \leq \|y_n - p\|^2 - \bigg(1 - \mu^2\frac{s^2_n}{s^2_{n+1}}\bigg)\|y_n - v_n\|^2.
		\end{equation}


\noindent On the other hand, one can see that \eqref{c^2} follows from \eqref{csd1}.	


\end{proof}
	
	
	
	\begin{remark}\label{Rem4.3}
	\noindent By Lemma \ref{Lem4.1} and $\mu\in (0,1),$ there exists $n_0\in\mathbb{N}$ such that $1 - \mu^2\frac{s_n^2}{s_{n+1}^2}>\epsilon>0$ for all $n\ge n_0.$ Consequently, it follows from \eqref{c1} that for all $p\in \Gamma$ and $n\ge n_0$
	
	\begin{equation*}
		\|t_n - p\|^2 \leq \|y_n - p\|^2 - \epsilon\|y_n - v_n\|^2.
	\end{equation*}
\end{remark}	
	

	
	
	

\begin{theorem}\label{the3}
		\noindent Let $C$ and $Q$ be nonempty closed convex subsets of real Hilbert spaces $H_1$ and $H_2$, respectively. Let $A : H_1 \rightarrow H_2$ be a bounded linear operator, and let $\{S_i\}$ be a countable family of $k_i$-strictly pseudo-contractive multivalued mappings of $C$ into $CB(C).$ Let $F_1, \phi_1 : C \times C \rightarrow \mathbb{R}, F_2, \phi_2 : Q \times Q \rightarrow \mathbb{R} $ be bifunctions satisfying Assumptions \ref{Ass 1}. Suppose $\phi_1, \phi_2$ are monotone, $\phi_1$ is upper hemicontinuous, and $F_2$ and $\phi_2$ are upper semicontinuous in the first argument. Let $B : H_1 \rightarrow H_1$ be an $L-$Lipschitz continuous monotone mapping  and $D : H_1 \rightarrow 2^{H_1}$ be a maximal monotone operator such that $\Gamma = SGEP(F_1, \phi_1, F_2, \phi_2) \cap \bigcap_{i=1}^mF(S_i) \cap \Omega \neq \emptyset$, where $\Omega = (B + D)^{-1}(0)$ and $S_ip = \{p\}$ for each $p \in \cap_{i=1}^mF(S_i)$. Let $\{x_n\}$ be a sequence generated by Algorithm \ref{alg3a} such that conditions (C1) and (C2) hold. Then, the sequence $\{x_n\}$ converges strongly to $q = P_{\Gamma}x_0.$
	\end{theorem}	
	
	
	\begin{proof}
		\noindent We divide the proof of the strong convergence Theorem \ref{the3} into various steps as follows:	 	
		
		\noindent \underline{Step 1:} We show that sequence $\{x_n\}$ generated by Algorithm \ref{alg3a} is bounded and well defined.
		
		\noindent Let $p \in \Gamma,$ then we have $p = T_{r_n}^{(F_1, \phi_1)}p$ and $Ap = T_{r_n}^{(F_1, \phi_1)}Ap, S_ip = p,$ for all $i = 1, 2, ...m.$
		
		\noindent Since $T_{r_n}^{(F_1, \phi_1)}$ is nonexpansive, then by Lemma \ref{lem3f} we have
		\begin{align}
			\|z_n - p\|^2 & = \|T_{r_n}^{(F_1, \phi_1)}(w_n - \gamma_nA^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n) - p\|^2\nonumber\\
			& \leq \|w_n - \gamma_nA^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n - p\|^2\nonumber\\
			&= \|w_n - p\|^2 + \gamma_n^2\|A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2 \nonumber\\
	       &- 2\gamma_n\langle w_n - p, A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\rangle. \label{c5}
		\end{align}
		By the firmly nonexpansivity of $I - T_{r_n}^{(F_2, \phi_2)}$, we get
		\begin{align}
			\langle w_n - p, A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n \rangle & = \langle Aw_n - Ap, (I - T_{r_n}^{(F_2, \phi_2)})Aw_n \rangle\nonumber\\
			& = \langle Aw_n - Ap, (I - T_{r_n}^{(F_2, \phi_2)})Aw_n \nonumber\\
	       &- (I - T_{r_n}^{(F_2, \phi_2)})Ap \rangle\nonumber\\
			& \geq \|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2.\label{c6}
		\end{align}
		By substituting \eqref{c6} in \eqref{c5} and applying the condition on $\tau_n,$ we have
		\begin{align}
			\|z_n - p\|^2 & \le \|w_n - p\|^2 + \gamma_n^2\|A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2 \nonumber\\
	        &- 2\gamma_n\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2\nonumber\\
			& = \|w_n - p\|^2 - \gamma_n\bigl[2\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2 \nonumber\\
	       &- \gamma_n\|A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2\bigr]\nonumber\\
			& = \|w_n - p\|^2 - \gamma_n(2 - \tau_n)\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2\label{cx}\\
			& \leq \|w_n - p\|^2.\label{c7}
		\end{align}
		
		\noindent By Lemma \ref{lem3b} and applying the fact that $S_i,~i=1,2,\ldots,m$ is strictly pseudo-contractive together with condition (C2), we get
		\begin{align}
			\|y_n - p\|^2 & = \|\alpha_{n, 0}z_n + \sum_{i=1}^{m}\alpha_{n, i}u_{n, i} - p\|^2\nonumber\\
			& = \alpha_{n, 0}\|z_n - p\|^2 + \sum_{i=1}^{m}\alpha_{n, i}\|u_{n, i} - p\|^2 \nonumber\\
	       &- \sum_{i=1}^{m}\alpha_{n, 0}\alpha_{n, i}\|u_{n, i} - z_n\|^2 - \sum_{1 \leq i < j\le m}\alpha_{n,i}\alpha_{n,j}\|u_{n, i} - u_{n, j}\|^2\nonumber\\
			& \leq \alpha_{n, 0}\|z_n - p\|^2 + \sum_{i=1}^{m} \alpha_{n,i}\big(H(S_iz_n, S_ip)\big)^2 \nonumber\\
            &- \sum_{i=1}^{m}\alpha_{n, 0}\alpha_{n, i}\|u_{n, i} - z_n\|^2 - \sum_{1 \leq i < j\le m}\alpha_{n, i}\alpha_{n, j}\|u_{n, i} - u_{n, j}\|^2\nonumber\\
			& \leq \alpha_{n, 0}\|z_n - p\|^2 + \sum_{i=1}^{m}\alpha_{n, i}\bigl(\|z_n - p\|^2 + k_i\|u_{n, i} - z_n\|^2\bigr) \nonumber\\
	       &- \sum_{i=1}^{m}\alpha_{n, 0}\alpha_{n, i}\|u_{n, i} - z_n\|^2\nonumber\\
			&~~ - \sum_{1 \leq i < j\le m}\alpha_{n, i}\alpha_{n, j}\|u_{n, i} - u_{n, j}\|^2\nonumber\\
			& \leq \|z_n - p\|^2 - \sum_{i=1}^{m}\alpha_{n, i}(\alpha_{n, 0} - k_i)\|u_{n, i} - z_n\|^2 \label{c2}\\
			& \leq \|z_n - p\|^2, \label{c3}
		\end{align}
		
		\noindent which implies that
		\begin{equation}\label{c4}
			\|y_n - p\| \leq \|z_n - p\|.
		\end{equation}
		
		\noindent By applying \eqref{c3} and \eqref{c7} into \eqref{lgi}, we get
		
		\begin{equation}\label{lmg}
			\|t_n - p\|^2  \leq \|w_n - p\|^2 - \bigg(1 - \mu^2\frac{s^2_n}{s^2_{n+1}}\bigg)\|y_n - v_n\|^2, \forall p \in \Gamma.
		\end{equation}
		
		\noindent By Lemma \ref{Lem2.4}, we have that $C_{n+1}$ is closed and convex. Furthermore, from \eqref{lmg} it follows that $p \in C_{n+1}.$ Hence, we have $\Gamma \subset C_{n+1} \subset C_n$ for all $n$ and thus $x_{n+1} = P_{C_{n+1}}x_0$ is well defined. Therefore, $\{x_n\}$ is well defined.
		
		\noindent We now show that $\{x_n\}$ is bounded.
		\noindent It is known that $\Gamma$ is a nonempty closed convex subset of $H_1$, then there exists a unique $q \in \Gamma$ such that $q = P_{\Gamma}x_0$. From $x_n = P_{C_n}x_0$ and $x_{n+1} \in C_{n+1}$ for all $n \in \mathbb{N},$ we obtain
		\begin{equation*}
			\|x_n - x_0\| \leq \|x_{n+1} - x_0\|,~~\forall~~n \in \mathbb{N}.
		\end{equation*}
		On the other hand, since $\Gamma \subset C_n,$ we get
		\begin{equation*}
			\|x_n - x_0\| \leq \|q - x_0\|,~~\forall~n \in \mathbb{N}.
		\end{equation*}
		This implies that $\{\|x_n - x_0\|\}$ is bounded. Hence, $\{x_n\}$ is bounded. Consequently $\{w_n\}, \{t_n\}, \{z_n\}$ and $\{y_n\}$ are bounded. Thus, $\lim_{n\to \infty}\|x_n - x_0\|$ exists.
		
		
		\noindent \underline{Step 2:} We claim that $\lim_{n\to \infty}x_n = q$, for some $q \in C.$
		
		\noindent It is clear from the definition of $C_n$ that $x_m = P_{C_m}x_0 \in C_m \subset C_n$, $m > n \geq 1.$ By Lemma \ref{lem3a1}, we obtain
		\begin{equation}\label{c11}
			\|x_m - x_n\|^2 \leq \|x_m - x_0\|^2 - \|x_n - x_0\|^2.
		\end{equation}
		Since $\lim_{n\to \infty}\|x_n - x_0\|$ exists, then it follows from \eqref{c11} that $\|x_m - x_n\| \rightarrow 0$ as $n \rightarrow \infty.$ Thus, $\{x_n\}$ is a Cauchy sequence. Since $H_1$ is complete and $C$ is closed, there exists $q\in C$ such that $x_n \rightarrow q $ as $n \to \infty.$
		
		
		\noindent \underline{Step 3:} We now show that $q\in\Gamma.$
		
		\noindent From \eqref{c11}, we obtain
		\begin{equation}\label{c12}
			\lim_{n\to \infty}\|x_{n+1} - x_n\| = 0.
		\end{equation}
		
		\noindent From the definition of $w_n$ and by applying  \eqref{c12}, we get
		\begin{align}\label{c15}
			\|w_n - x_n\| = |\theta_n|\|x_n - x_{n-1}\|\le |\theta|\|x_n - x_{n-1}\|\to 0,\quad n\to\infty .
		\end{align}
		
		\noindent From \eqref{c12} and \eqref{c15}, we obtain
		
		\begin{align}\label{Eq4.15}
			\|w_n - x_{n+1}\| \to 0,\quad n\to\infty .
		\end{align}
		
		\noindent We known that $x_{n+1} \in C_{n+1}.$ Then, from the definition of $C_{n+1}$ we obtain
		
		\begin{equation*}
			\|t_n - x_{n+1}\|^2 \leq \|w_n - x_{n+1}\|^2.
		\end{equation*}
		
		\noindent Combining this with \eqref{Eq4.15}
		gives
		
		\begin{equation}\label{c13}
			\lim_{n\to \infty}\|t_n - x_{n+1}\| = 0.
		\end{equation}
		
		\noindent From \eqref{c12} and \eqref{c13}, we obtain
		
		\begin{equation}\label{c14}
			\lim_{n\to \infty}\|t_n - x_n\| = 0.
		\end{equation}
		
		
		
		\noindent From \eqref{c15} and \eqref{c14}, we obtain
		\begin{equation}\label{c16}
			\lim_{n\to \infty}\|t_n - w_n\| = 0.
		\end{equation}
		
		
		\noindent By applying \eqref{c3} and \eqref{c7} into Remark \ref{Rem4.3}, we have

\begin{equation*}
\|t_n - p\|^2 \leq \|w_n - p\|^2 - \epsilon\|y_n - v_n\|^2.
\end{equation*}

\noindent From which we get	
		\begin{align*}
			\epsilon\|y_n - v_n\|^2	&\leq \|w_n - p\|^2 - \|t_n - p\|^2\\
			&\leq \|w_n - t_n\|(\|w_n - p\| + \|t_n - p\|),
		\end{align*}
		
		
		\noindent which together with \eqref{c16} implies that
		
		
		\begin{align}\label{Eq4.19}
			\|y_n - v_n\|\to 0,\quad n\to\infty.
		\end{align}
		
		\noindent Applying Lemma \ref{Lem4.1} together with \eqref{Eq4.19} to \eqref{c^2}, we have
		
		
		
		\begin{align}\label{Eq4.20}
			\|t_n - v_n\|\to 0,\quad n\to\infty.
		\end{align}
		
		\noindent From \eqref{c16}-\eqref{Eq4.20}, we obtain
		
		
		\begin{align}\label{Eq4.21}
			\|y_n - w_n\|\to 0,\quad n\to\infty.
		\end{align}
		
		
		\noindent From \eqref{c7} and \eqref{c2}, we obtain
		
		\begin{equation*}
			\|y_n - p\|^2\leq \|w_n - p\|^2 - \sum_{i=1}^{m}\alpha_{n, i}(\alpha_{n, 0} - k_i)\|u_{n, i} - z_n\|^2.
		\end{equation*}
		
		\noindent From this we have
		
		\begin{align*}
			\alpha_{n, i}(\alpha_{n, 0} - k_i)\|u_{n, i} - z_n\|^2 &\le\sum_{i=1}^{m}\alpha_{n, i}(\alpha_{n, 0} - k_i)\|u_{n, i} - z_n\|^2\\
			&\le\|w_n - p\|^2-\|y_n - p\|^2\\
			&\le (\|w_n - y_n\|)(\|w_n - p\|+ \|y_n - p\|).
		\end{align*}
		
		\noindent By applying Condition (C2) and \eqref{Eq4.21}, we get
		
		
		\begin{align}\label{Eq4.26}
			\|u_{n, i} - z_n\|\to 0,\quad n\to\infty.
		\end{align}
		
		
		\noindent From the definition of $y_n$ and by applying \eqref{Eq4.26}, we get
		
		
		
		\begin{equation}\label{Eq4.23}
			\|y_n - z_n\| \le \alpha_{n, 0}\|z_n-z_n\| + \sum_{i=1}^{m}\alpha_{n, i}\|u_{n, i} - z_n\|\to 0,\quad n\to\infty.
		\end{equation}
		
		
		
		
		
		
		
		\noindent Also, by applying \eqref{c15}, \eqref{Eq4.21} and \eqref{Eq4.23}, we obtain
		\begin{equation}\label{cxh}
			\lim_{n \to \infty}\|w_n - z_n\|= 0;\quad
			\lim_{n \to \infty}\|z_n - x_n\| =0 .
		\end{equation}
		
		\noindent From \eqref{cx}, we have
		\begin{equation*}
			\|z_n - p\|^2 \leq \|w_n - p\|^2 - \gamma_n(2 - \gamma_n)\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2,
		\end{equation*}
		\noindent which implies that
		
		\begin{align*}
			\gamma_n(2 - \gamma_n)\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2 & \leq \|w_n - p\|^2 - \|z_n - p\|^2\\
			& \leq \|w_n - z_n\|(\|w_n - p\| + \|z_n - p\|).
		\end{align*}
		Using the definition of $\gamma_n$, the condition on $\tau_n$ and applying \eqref{cxh}, it follows that
		\begin{equation*}
			\frac{\tau_n(2 - \tau_n)\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^4}{\|A^*(I - T_{r_n}^{(F_2, \phi_2)Aw_n})Aw_n\|^2} \rightarrow 0\quad\text{as}~~n \rightarrow \infty.
		\end{equation*}
		From which we get
		\begin{equation*}
			\frac{\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|^2}{\|A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|} \rightarrow 0\quad\text{as}~~n \rightarrow \infty.
		\end{equation*}
		
		
		\noindent Since $\|A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\|$ is bounded, then it follows that
		
		\begin{equation}\label{Eq4.24}
			\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n\| \rightarrow 0,\quad~~n \rightarrow \infty.
		\end{equation}
		
		\noindent Consequently, we have
		\begin{align}
			\|A^*(I - T_{r_n}^{(F_2, \phi_2)})Aw_n)\| & \leq \|A^*\|\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n)\|\nonumber\\
			&= \|A\|\|(I - T_{r_n}^{(F_2, \phi_2)})Aw_n)\|\rightarrow 0~~\text{as}~~n \rightarrow \infty.\label{c17}
		\end{align}
		
		\noindent Since $t_n = v_n - s_n(Bv_n - By_n)$ and $B$ is Lipschitz continuous, then by applying \eqref{Eq4.19} we have
		\begin{align*}
			\|t_n - v_n\|  = \|v_n - s_n(Bv_n - By_n) - v_n\| = s_n\|By_n - Bv_n\|\to 0,\quad n\to\infty.
		\end{align*}
		
		
		\noindent Since $\{x_n\}$ is bounded, then $w_\omega(x_n)$ is nonempty. Let $q\in w_\omega(x_n)$ be an arbitrary element. Then there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ such that $x_{n_k}\rightharpoonup q$ as $k\to\infty.$ Let $z\in w_\omega(x_n)$ and $\{x_{n_j}\}\subset \{x_n\}$ be such that $x_{n_j}\rightharpoonup z$ as $j\to\infty.$ From \eqref{cxh}, we get $z_{n_k}\rightharpoonup q$ and $z_{n_j}\rightharpoonup z.$
		Since $I-S_i$ is demiclosed at zero for each $i=1,2,\ldots,m,$ then it follows from \eqref{Eq4.26} that $q,z\in F(S_i)$ for all $i=1,2,\ldots,m,$ which implies that $q,z\in\cap_{i=1}^m F(S_i).$
		
		
		
		\noindent Next, let $(g, h) \in \text{Graph}(B + D),$ that is $h - Bg\in Dg.$ Since $v_{n_k} = (I + s_{n_k}D)^{-1}(I - s_{n_k}B)y_{n_k},$ we have
		
		\begin{equation*}
			(I - s_{n_k}B)y_{n_k} \in (I + s_{n_k}D)v_{n_k},
		\end{equation*}
		which implies that
		\begin{equation*}
			\frac{1}{s_{n_k}}(y_{n_k} - v_{n_k} - s_{n_k}By_{n_k}) \in Dv_{n_k}.
		\end{equation*}
		Since $D$ is maximal monotone, we get
		\begin{equation*}
			\biggl\langle g - v_{n_k}, h - Bg - \frac{1}{s_{n_k}}(y_{n_k} - v_{n_k} - s_{n_k}By_{n_k}) \biggr\rangle \geq 0.
		\end{equation*}
		
		
		\noindent From this we obtain
		
		\begin{align*}
			\langle g - v_{n_k}, h \rangle & \geq \biggl\langle g - v_{n_k}, Bg + \frac{1}{s_{n_k}}(y_{n_k} - v_{n_k} - s_{n_k}By_{n_k}) \biggr\rangle\\
			& = \langle g - v_{n_k}, Bg - By_{n_k} \rangle + \biggl\langle g - v_{n_k}, \frac{1}{s_{n_k}}(y_{n_k} - v_{n_k}) \biggr\rangle\\
			& = \langle g - v_{n_k}, Bg - Bv_{n_k} \rangle + \langle g - v_{n_k}, Bv_{n_k} - By_{n_k} \rangle \nonumber\\
	       &+ \biggl\langle g - v_{n_k}, \frac{1}{s_{n_k}}(y_{n_k} - v_{n_k}) \biggr\rangle\\
			& \geq \langle g - v_{n_k}, Bv_{n_k} - By_{n_k} \rangle + \biggl\langle g - v_{n_k}, \frac{1}{s_{n_k}}(y_{n_k} - v_{n_k}) \biggr\rangle.
		\end{align*}
		Since $B$ is Lipschitz continuous and $\lim_{n \to \infty}\|v_n - y_n\| = 0$, we have $\lim_{n \to \infty}\|Bv_{n_k} - By_{n_k}\| = 0.$ Applying this together with $\lim_{n\to \infty}s_n = s \geq \min\biggl\{s_1, \frac{\mu}{L}\biggr\}$, we get
		
		\begin{equation}\label{Eq4.27}
			\langle g - q, h \rangle = \lim_{k \to \infty}\langle g - v_{n_k}, h \rangle \geq 0.
		\end{equation}
		
		
		\noindent Following similar argument, we obtain
		
		\begin{equation}\label{Eq4.28}
			\langle g - z, h \rangle = \lim_{j \to \infty}\langle g - v_{n_j}, h \rangle \geq 0.
		\end{equation}
		
		
		\noindent By the maximal monotonicity of $(B + D)$, it follows from \eqref{Eq4.27} and \eqref{Eq4.28} that
		$q,z \in (B + D)^{-1}(0).$
		
		
		\noindent Next, since $z_{n_k} = T^{(F_1, \phi_1)}_{r_{n_k}}(I - \gamma_{n_k}A^*(I - T_{r_{n_k}}^{F_2, \phi_2})A)w_{n_k},$ then by applying Lemma \ref{lem3c}, we get
		\begin{align*}
			F_1(z_{n_k}, y) + \phi_1(z_{n_k}, y) & \nonumber\\
	& + \frac{1}{r_{n_k}}\langle y - z_{n_k}, z_{n_k} - w_{n_k} - \gamma_{n_k}A^*(I - T_{r_{n_k}}^{(F_2, \phi_2)})Aw_{n_k} \rangle \nonumber\\
	& \geq 0,~~\forall y \in C,
		\end{align*}
		which implies that
		\begin{align*}
			F_1(z_{n_k}, y) + \phi_1(z_{n_k}, y) &\\
&+ \frac{1}{r_{n_k}}\langle y - z_{n_k}, z_{n_k} - w_{n_k}\rangle &\nonumber\\
	 &-\frac{1}{r_{n_k}}\langle y - z_{n_k}, \gamma_{n_k}A^*(I - T_{r_{n_k}}^{(F_2, \phi_2)})Aw_{n_k} \rangle \nonumber\\
	&\geq 0,~~\forall y \in C.
		\end{align*}
		From the monotonicity of $F_1$ and $\phi_1$, it follows that
		\begin{align*}
			\frac{1}{r_{n_k}}\langle y - z_{n_k}, z_{n_k} - w_{n_k}\rangle & \nonumber\\
	&-\frac{1}{r_{n_k}}\langle y - z_{n_k}, \gamma_{n_k}A^*(I - T_{r_{n_k}}^{(F_2, \phi_2)})Aw_{n_k} \rangle\nonumber\\
	& \geq F_1(y, z_{n_k}) + \phi_1(y, z_{n_k}),~~\forall y \in C.
		\end{align*}
		\noindent By \eqref{cxh} and $x_{n_k} \rightharpoonup q$, we obtain $z_{n_k} \rightharpoonup q$. Applying condition $(C1)$, \eqref{cxh}, \eqref{c17} and Assumption \ref{Ass 1} (A1)-(A7), we obtain
		\begin{equation*}
			0 \geq F_1(y, q) + \phi_1(y, q),~~\forall y \in C.
		\end{equation*}
		Suppose $y_t = ty + (1 - t)q, \forall t \in (0, 1]$ and $y \in C.$ Then, $y_t \in C$ and $F_1(y_t, q) + \phi_1(y_t, q) \leq 0.$ Therefore, by Assumption \ref{Ass 1} (A1)-(A7), we get
		\begin{align*}
			0 & \leq F_1(y_t, y_t) + \phi_1(y_t, y_t)\\
			& \leq t\bigl(F_1(y_t, y) + \phi_1(y_t, y)\bigr) + (1 - t)\bigl(F_1(y_t, q) + \phi_1(y_t, q)\bigr)\\
			& \leq t\bigl(F_1(y_t, y) + \phi_1(y_t, y)\bigr).
		\end{align*}
		Thus, we have
		\begin{equation*}
			F_1(y_t, y) + \phi_1(y_t, y) \geq 0,~~\forall y \in C.
		\end{equation*}
		Letting $t \to 0,$ and applying condition (A3) together   with the upper hemicontinuity of $\phi_1,$ we have
		\begin{equation}\label{Eq4.29}
			F_1(q, y) + \phi_1(q, y) \geq 0,~~\forall y \in C.
		\end{equation}
		
		\noindent By similar argument, we have
		
		
		\begin{equation}\label{Eq4.30}
			F_1(z, y) + \phi_1(z, y) \geq 0,~~\forall y \in C.
		\end{equation}
		
		\noindent It follows from \eqref{Eq4.29} and \eqref{Eq4.30} that
		$q,z \in GEP(F_1, \phi_1)$.
		
		
		
		\noindent Next, we show that $Aq,Az \in GEP(F_2, \phi_2)$. Since $A$ is a bounded linear operator, then by \eqref{c15} we have $Aw_{n_k} \rightharpoonup Aq.$ Hence, from \eqref{Eq4.24}, we obtain
		\begin{equation}\label{c32}
			T_{r_{n_k}}^{(F_2, \phi_2)}Aw_{n_k} \rightharpoonup Aq,\quad k\to\infty.
		\end{equation}
		By the definition of $T_{r_{n_k}}^{(F_2, \phi_2)}Aw_{n_k}$, we have
		\begin{align*}
			F_2(T_{r_{n_k}}^{(F_2, \phi_2)}Aw_{n_k}, y) & \nonumber\\
	&+ \phi_2(T_{r_{n_k}}^{(F_2, \phi_2)}Aw_{n_k}, y) \nonumber\\
	&+ \frac{1}{r_{n_k}}\langle y - T_{r_{n_k}}^{(F_2, \phi_2)}Aw_{n_k}, T_{r_{n_k}}^{(F_2, \phi_2)}Aw_{n_k} - Aw_{n_k} \rangle \nonumber\\
	&\geq 0,~~\forall y \in Q.
		\end{align*}
		Since $F_2$ and $\phi_2$ are upper semicontinuous in the first argument, then by \eqref{Eq4.24}, \eqref{c32} and $\liminf_{k\to \infty}r_{n_k} > 0,$ we have
		
		\begin{equation}\label{Eq4.32}
			F_2(Aq, y) + \phi_2(Aq, y) \geq 0,~~\forall y \in Q.
		\end{equation}
		
		
		\noindent Following similar argument, we have
		
		\begin{equation}\label{Eq4.33}
			F_2(Az, y) + \phi_2(Az, y) \geq 0,~~\forall y \in Q.
		\end{equation}
		
		\noindent From \eqref{Eq4.32} and \eqref{Eq4.33}, it follows that $Aq,Az \in GEP(F_2, \phi_2).$ Therefore $q,z \in SGEP(F_1, \phi_1, F_2, \phi_2).$   By Invoking Lemma \ref{lem3g}, we get $q=z.$ Hence, we have that $q\in\Gamma.$
		
		
		
		\noindent \underline{Step 4.} Lastly, we show that $q = P_{\Gamma}x_0.$
		
		\noindent\\
		\noindent Since $x_n = P_{C_n}x_0$ and $\Gamma \subset C_n,$  we have $\langle x_0 - x_n, x_n - p \rangle \geq 0$ for all $p \in \Gamma$. By taking limit as $n\to\infty,$ we have $\langle x_0 - q, q - p \rangle \geq 0$ for all $p \in \Gamma$. This shows that $q = P_{\Gamma}x_0$.
		
		\noindent Therefore, we can conclude by the steps above that $\{x_n\}$ converges strongly to $q = P_{\Gamma}x_0$. This completes the proof.
	\end{proof}
	
	\noindent If $\phi_1 = \phi_2 = 0$ in \eqref{c3x}-\eqref{c4x}, then the split generalized equilibrium problem reduces to split equilibrium problem. Hence from Theorem \ref{alg3a} , we obtain the following consequent result.
	
	\begin{corollary}\label{corl1x}
		\noindent Let $C$ and $Q$ be nonempty closed convex subsets of real Hilbert spaces $H_1$ and $H_2$, respectively. Let $A : H_1 \rightarrow H_2$ be a bounded linear operator, and let $\{S_i\}_{i=1}^m$ be a countable family of $k_i$-strictly pseudo-contractive multivalued mappings of $C$ into $CB(C)$ such that $I-S_i$ is demiclosed at zero for each $i=1,2,\ldots,m,$  $S_ip = \{p\}$ for each $p \in \cap_{i=1}^mF(S_i)$ and $k=\max\{k_i\}.$ Let $F_1: C \times C \rightarrow \mathbb{R}, F_2: Q \times Q \rightarrow \mathbb{R}$ be bifunctions satisfying Assumptions \ref{Ass 1} such that $F_2$ is upper semicontinuous in the first argument. Let $B : H_1 \rightarrow H_1$ be $L$-Lipschitz continuous and monotone and $D : H_1 \rightarrow 2^{H_1}$ be a maximal monotone operator such that $\Gamma = SGEP(F_1, F_2)  \cap_{i=1}^mF(S_i) \bigcap (B + D)^{-1}(0) \neq \emptyset.$ Let $\{x_n\}$ be a sequence generated as follows:
		
\hrule		
		\begin{algorithm}\label{alg4a}\noindent
			\hrule \hrule
			
			\noindent \textbf{Initialization:} Select $x_0, x_1 \in H_1$, $\mu \in (0, 1), \theta_n \in [-\theta, \theta]$ for some $\theta>0$ and $C_1=C.$
			
			\noindent \textbf{Iterative Step:} Given the current iterate $x_n$, calculate the next iterate as follows:
			
			\noindent \textbf{Step 1 :} Compute
			\begin{equation*}
				w_n = x_n + \theta_n(x_n - x_{n - 1}).
			\end{equation*}
			\noindent \textbf{Step 2 :} Compute
			\begin{equation*}
				z_n = T_{r_n}^{F_1}(I - \gamma_nA^*(I - T_{r_n}^{F_2})A)w_n.
			\end{equation*}
			\noindent \textbf{Step 3 :} Compute
			\begin{equation*}
				y_n = \alpha_{n, 0}z_n + \sum_{i = 1}^{m}\alpha_{n, i}u_{n, i},\quad u_{n, i} \in S_iz_n.
			\end{equation*}
			\noindent \textbf{Step 4 :} Compute
			\begin{equation*}
				\begin{cases}
					v_n = (I + s_nD)^{-1}(I - s_nB)y_n \\
					t_n = v_n - s_n(Bv_n - By_n)\\
					C_{n+1} = \{p \in C_n : \|t_n - p\|^2 \leq \|x_n - p\|^2 - 2\theta_n\langle
					x_n - p, x_{n-1} - x_n\rangle \\
                    + \theta_n^2\|x_{n-1} - x_n\|^2\}\\
					x_{n+1} = P_{C_{n+1}}x_0,
				\end{cases}
			\end{equation*}
			\noindent \textbf{Step 5 :} Compute
			\begin{equation}
				s_{n+1} = \begin{cases}
					\min\biggl\{\frac{\mu\|y_n - v_n\|}{\|By_n - Bv_n\|} , s_n\biggr\} & \text{if}~~By_n - Bv_n \neq 0.\\
					s_n   & \text{otherwise},
				\end{cases}
			\end{equation}
			\noindent Set $n := n + 1$ and return to {\bf Step 1}.
			\noindent where
			\begin{equation*}
				\gamma_n =
				\begin{cases}
					\tau_n\frac{||(I - T_{r_n}^{(F_2})Aw_n||^2}{||A^*(I - T_{r_n}^{(F_2})Aw_n||^2} &  \text{If~~} Aw_n \neq T^{(F_2}_{r_n}Aw_n\\
					
					\gamma  &  \text{otherwise}~~(\gamma~~\text{being any non-negative real number.})
				\end{cases}
			\end{equation*}
			
		\end{algorithm}
		\hrule
		
		
		\noindent Suppose other conditions of Theorem \ref{alg3a} hold. Then, the sequence $\{x_n\}$ converges strongly to $q = P_{\Gamma}x_0.$
	\end{corollary}
	
	
	
	
	
	
	
	\section{Applications}\label{Sec5}
	
	
	\subsection{Split Minimization Problem}
	\noindent \\ Let $H_1,~ H_2$ be two real Hilbert spaces, and let $C\subset H_1$ and $Q\subset H_2$ be nonempty, closed, and convex subsets. Let $f:C\to\mathbb{R}, ~~ ~~ g:Q\to\mathbb{R}$ be two operators and $A:H_1\to H_2$ be a bounded linear operator. The split minimization problem (SMP) is formulated as finding
	\begin{equation}\label{Eq5.1B}
		x^*\in C \quad \text{such that}\quad f(x^*)\le f(x), ~~ ~ \forall x\in C,
	\end{equation}
	
	and
	
	\begin{equation}\label{Eq5.2A}
		y^*=Ax^* \quad \text{such that}\quad g(y^*)\le g(y), ~~ ~ y\in Q.
	\end{equation}
	\noindent Let $\Omega$ denote the set of solution of SMP \eqref{Eq5.1B}-\eqref{Eq5.2A}, and we assume $\Omega \ne \emptyset.$ Let $\phi_1=\phi_2=0,$ and
	$$F_1(x,y):= f(y)-f(x)~~ ~~\quad \text{for all}~~ ~ x,y\in C;$$
	\noindent and
	$$F_2(u,v):= g(v)-g(u)\quad \text{for all}~~ ~ u,v\in Q.$$
	
	\noindent Suppose $f$ and $g$ are convex and lower semi-continuous on $C$ and $Q,$ respectively. Then, $F_1, F_2, \phi_1$ and $\phi_2$ satisfy all the conditions of Assumption \ref{Ass 1}. Consequently, from Theorem \ref{alg3a} we obtain a strong convergence theorem for approximating a common solution of split minimization problem, monotone variational inclusion problem and fixed point problem for a countable family of strict pseudo-contractive multivalued mappings in Hilbert spaces.
	
	
	
	\subsection{Split Variational Inequality Problem}
	\noindent \\ Let $C$ be a nonempty closed convex subset of a real Hilbert space $H$, and $f:H\to H$ be a single-valued mapping. The variational inequality problem (VIP) introduced independently by Fichera \cite{FicG} and Stampacchia \cite{StampG} is formulated as follows:
	\begin{align} \label{Eq5.2}
		\text{find}~~ x^* \in C ~~ \text{such that}~~ \langle y-x^*, fx^* \rangle \ge 0, \quad \quad \forall~ y\in C.
	\end{align}
	\noindent The VIP can be modelled to solve several optimization problems and has vast applications  in different fields, such as in physics, engineering,  economics, etc, (see \cite{alakoya2, BlOe, CeGiRe, dph, ogwo, ogwo3, sitt}).
	
	\noindent The split variational inequality problem (SVIP), which was first introduced by Censor et al. \cite{CeGiRe} is defined as finding a point:\
	\begin{align}
		& x^* \in C ~~ \text{such that}~~ \langle x-x^*, f(x^*) \rangle \ge 0 \quad \quad \forall~ x\in C,\label{Eq5.3}\\ \text{and}\quad&\nonumber\\
		& y^* =Ax^*\in Q ~~ \text{solves}~~ \langle y-y^*, g(y^*) \rangle \ge 0 \quad \quad \forall~ y\in Q,\label{Eq5.4}
	\end{align}
	\noindent where $C$ and $Q$ are nonempty, closed, convex subsets of real Hilbert spaces $H_1$ and $H_2,$  respectively, $f:H_1\to H_1$ and $g:H_2\to H_2$ are monotone mappings, and $A:H_1\to H_2$ is a bounded linear operator, see \cite{LoKaF}. Let $\Omega \ne \emptyset$ denote the set of solution of SVIP \eqref{Eq5.3}-\eqref{Eq5.4}. By setting $\phi_1=\phi_2=0,$ and
	$$F_1(x,y):= \langle y-x, f(x) \rangle \quad \text{for all}~~ ~ x,y\in C;$$
	\noindent\hspace{110pt} and
	$$F_2(u,v):= \langle v-u, g(u) \rangle \quad \text{for all}~~ ~ u,v\in Q.$$
	\noindent Then, $F_1, F_2, \phi_1$ and $\phi_2$ satisfy all the conditions of Assumption \ref{Ass 1}. Hence, from Theorem \ref{alg3a}, we obtain a strong convergence theorem for approximating a common solution of split variational inequality problem, monotone variational inclusion problem and fixed point problem for a countable family of strict pseudo-contractive multivalued mappings in Hilbert spaces.
	
	
	
	
	
	
	
	
	
	
	
	
	
	
	
	
	
	
	
	\section{Numerical Examples}\label{Sec6}
	
	\noindent In this section, we present a numerical experiments to illustrate the
	performance of our Algorithm \ref{alg3a} as well as comparing it with Algorithm \eqref{algdep}, Algorithm \eqref{algsitth}, Algorithm \eqref{algphu} and Algorithm \eqref{xii} in the literature.
	
	
	
	\noindent In our computation,  we choose $\alpha_{n,0}=\frac{n}{2n+1}, \alpha_{n,i}=\frac{n+1}{5(2n+1)},~i=1,2,\ldots,5, \tau_n=1.5, \theta_n=1.9, r_n=2.0, s_0=0.1$ and $\mu=0.7$ in our Algorithm \ref{alg3a}. $Gx=\frac{1}{3}x,fx=\frac{2}{3}x,Kx=\frac{2}{5}x, \lambda_n=\frac{2n}{5n+1}, \alpha_n=\frac{2}{2n+3},\beta_n =\frac{n+1}{2n+3}, \eta=\frac{2}{5},\gamma=0.2, T_jx=\frac{2}{(3+j)}x$ in Algorithm \eqref{algdep},$ \beta_n=\xi_n=\frac{1}{2}(1-\alpha_n),\sigma_n=\frac{2}{2n+1}$ in Algorithm \eqref{algsitth} while in
	Algorithm \eqref{algphu} and Algorithm \eqref{xii}. Let the sequences $\{\delta_{n,j}\}$ be defined as follows for each $j\in\mathbb{N}\cup\{0\}$ and $n\in\mathbb{N}:$
	\begin{equation}\label{Eq5.1A}
		\delta_{n,j} =\begin{cases}
			\frac{1}{b^{j+1}}(\frac{n}{n+1}),\hspace{67pt} n> j,\\
			1-\frac{n}{n+1}(\sum_{k=1}^{n}\frac{1}{b^k}),\hspace{30pt} n =j,\\
			0, \hspace{105pt} n<j,
		\end{cases}
	\end{equation}
	\noindent where $b>1.$
	
	
	
	
	\begin{example}\label{EX1}
		Let $H_1=H_2=\mathbb{R}$ and $C=Q=[0,10].$ Let $A:H_1\to H_2$ be defined by $Ax=\frac{x}{5}$ for all $x\in H_1.$ Then, we have that $A^*y=\frac{y}{5}$ for all $y\in H_2.$ For $x\in C,j\in\mathbb{N}~~$and $ i=1,2,\ldots,5,$ let $P_j,S_i:C\to CB(C)$ be multivalued mappings defined as follows:
		\begin{equation}
			P_j(x) = \bigg[0, \frac{x}{10j}\bigg],\quad
			S_i(x) = \bigg[0, \frac{x}{10i}\bigg] .
		\end{equation}	
		\noindent One can easily verify that $P_j$ and $S_i$ are nonexpansive and strictly pseudo-contractive, respectively.  Define mappings $B:H_1\to H_1$ by $Bx=2x,$ $D:H_1\to H_1$ by $Dx=3x,$ and	
		let the bifunctions $F_1,\phi_1:C\times C\to\mathbb{R}$ be defined by $F_1(x,y)=y^2+3xy-4x^2$ and $\phi_1(x,y)=y^2-x^2$ for $x,y\in C,$ and $F_2,\phi_2:Q\times Q\to\mathbb{R}$ by $F_2(w,v)=2v^2+wv-3w^2$ and $\phi_2(w,v)=w-v$ for $w,v\in Q.$ It is easy to verify that all the conditions of Theorem \ref{the3} are satisfied. Next, we compute $T_r^{(F_1,\phi_1)}(x).$ We find $u\in C$ such that for all $z\in C$
		\begin{align*}
			0&\leq F_1(u,z) + \phi_1(u,z) + \frac{1}{r}\langle z - u, u - x \rangle \\
			&= 2z^2 + 3uz - 5u^2 + \frac{1}{r}\langle z - u, u - x \rangle\\
			&\Leftrightarrow\\
			0&\leq 2rz^2 + 3ruz - 5ru^2 + (z-u)(u-x)\\
			&= 2rz^2 + 3ruz - 5ru^2 + uz - xz - u^2 + ux\\
			&= 2rz^2 + (3ru+u-x)z + (-5ru^2 - u^2 + ux).
		\end{align*}
		\noindent Suppose $h(z) = 2rz^2 + (3ru+u-x)z + (-5ru^2 - u^2 + ux).$ Then, $h(z)$ is a quadratic function of $z$ with coefficients $a=2r, b=3ru+u-x,$ and $c=-5ru^2 - u^2 + ux.$ We determine the discriminant $\triangle$ of $h(z)$ as follows:
		\begin{align}
			\triangle &= (3ru+u-x)^2 -4(2r)(-5ru^2 - u^2 + ux)\nonumber\\
			&= 49r^2u^2 + 14ru^2 - 14rux + u^2 - 2ux + x^2\nonumber\\
			&= ((7r+1)u-x)^2.\label{cdx}
		\end{align}
		\noindent By Lemma \ref{lem3c},  $T_r^{(F_1,\phi_1)}$ is single-valued. Thus, it follows that $h(z)$ has at most one solution in $\mathbb{R}.$ Hence, from \eqref{cdx}, we have that $u= \frac{y}{7r+1}.$ This implies that $T_r^{(F_1,\phi_1)}(y)=\frac{y}{7r+1}.$ Similarly, we compute $T^{(F_2, \phi_2)}_r (y).$ Find $w\in Q$ such that for all $d\in Q$
		\begin{equation*}
			T^{(F_2, \phi_2)}_r (y) = \left\{ w \in Q : F_2 (w,d) + \phi_2(w,d) + \frac{1}{r} \langle d - w, w - y \rangle\ge 0,\quad \forall~ d \in Q \right\}.
		\end{equation*}
		By following similar procedure as above, we obtain $w=\frac{y+r}{5r+1}.$ This implies that $T^{(F_2, \phi_2)}_r (y)=\frac{y+r}{5r+1}.$
		
		\noindent In this example, we set the parameter $b$ on $\{\delta_{n,i}\}$ in \eqref{Eq5.1A} to be $b=40, v=3.5$ and we choose different initial values as follows:\\
		\noindent {\it Case I:} $x_0 = 7, x_1 = 3;$\\
		\noindent {\it Case II:} $x_0 = 6 , x_1 = 2;$\\
		\noindent {\it Case III:} $x_0 = 8, x_1 = 4;$\\
		\noindent {\it Case IV:} $x_0 = 9, x_1 = 5.$\\\\
		\noindent We compare the performance of our Algorithm \ref{alg3a} with Algorithms \eqref{algdep}, \eqref{algsitth},  \eqref{algphu} and \eqref{xii}. The stopping criterion used for our computation is $|x_{n+1}-x_{n}|< 10^{-4}$. We plot the graphs of errors against the number of iterations in each case.  The numerical results are reported in Figure \ref{fig1} and Table \ref{tab1}.
	
\begin{table}[h]
		\caption{\bf Numerical results for Example \ref{EX1}}
		\label{tab1}
		\begin{tabular}{ |p{1.3cm} |p{2.5cm}| p{1.2cm}| p{1.2cm}| p{1.2cm}| p{1.2cm}|  p{1.2cm}|}
			\hline
			\noindent & \noindent & Alg. \eqref{algdep} & Alg. \eqref{algsitth} & Alg. \eqref{algphu} & Alg. \eqref{xii} & Alg. \ref{alg3a}\\
			\hline
			Case I  & No. of Iter.  & 9 & 20 & 4 & 9 & 2\\
			&   CPU time (sec)  & 0.0057 & 0.0078 & 1.6693 & 0.3383 & 0.0032\\
			\hline
			Case II & No. of Iter. & 8 & 20 & 4 & 8 & 2\\
			&   CPU time (sec)  & 0.0051 & 0.0059 & 1.6884 & 0.3124 & 0.0039\\
			\hline
			Case III  & No. of Iter. & 9 & 20 & 4 & 9 & 2\\
			&   CPU time (sec)  & 0.0053 & 0.0057 & 1.6625 & 0.3566 & 0.0041\\
			\hline
			Case IV & No. of Iter. & 9 & 20 & 4 & 9 & 2 \\
			&   CPU time (sec)  & 0.0054 & 0.0067 & 1.6623 & 0.3449 & 0.0039\\
			\hline

			
			
		\end{tabular}
	\end{table}
	
	
	\begin{figure}
		\begin{center}
			\includegraphics[height=4cm]{Case1.pdf}	
			\includegraphics[height=4cm]{Case2.pdf}\\
			\includegraphics[height=4cm]{Case3.pdf}	
			\includegraphics[height=4cm]{Case4.pdf}\\
		\end{center}
		\caption{ Top left: Case I ; Top right: Case II; Bottom left:  Case III ; Bottom right: Case IV.}\label{fig1}
	\end{figure}
	
		

	\end{example}
	
	
	\begin{example}\label{EX2}
		Let $H_1=H_2 = L_2([0,1])$  with the inner product defined as
		\begin{equation*}
			\langle x, y \rangle = \int_0^1 x(t)y(t)dt, \quad \forall x,y\in L_2([0,1]).
		\end{equation*}
		\noindent Let
		$$C:=\{x\in H_1:\langle a,x \rangle \ge d\},$$
		\noindent where $a=2t^2$ and $d= 0.$ Here, we have
		$$P_C(x) = x + \frac{d-\langle a,x \rangle}{||a||^2}a.$$
		\noindent Also, let
		$$Q:=\{x\in H_2:\langle c,x \rangle\leq e\},$$
		\noindent where $c=\frac{t}{3},$  $e=1$ and we have
		$$P_Q(x) = x + \max \bigg\{0, \frac{e-\langle c,x \rangle}{||c||^2}c\bigg\}.$$
		\noindent Let $F_1:C\times C\rightarrow\mathbb{R}$ and $F_2:Q\times Q \rightarrow\mathbb{R}$ be defined as  $F_1(x,y) =\langle L_1x, y-x \rangle$ and $F_2(x,y) =\langle L_2x, y-x \rangle,$ where $L_1x(t) =\frac{x(t)}{3}$ and $L_2x(t) =\frac{x(t)}{4}.$ It can easily be verified that $F_1$ and $F_2$ satisfy conditions (A1)-(A4). Also, let $\phi_1=\phi_2=0.$ Furthermore, define $B:H_1\to H_1$ by $Bx=3x,$ $D:H_1\to H_1$ by $Dx=7x,$ and	let $A:L_2([0,1])\rightarrow L_2([0,1])$ be defined by $Ax(t)=\frac{x(t)}{3}$ and $A^*y(t)=\frac{y(t)}{3}.$  Then, $A$ is a bounded linear operator. We consider the case for which the multivalued mappings $\{S_j\}$ and $\{S_i\}$ are single-valued.	Let $S_j,S_i: L^2([0,1])\rightarrow L^2([0,1])$ be defined by
		$$
		(S_jx)(t) = \int_0^1 t^jx(s)ds\quad\quad\text{and}\quad\quad (S_ix)(t) = \int_0^1 t^ix(s)ds\;\;\;\text{for all}\,\, t\in [0,1].
		$$
		Note that $S_i$ and $S_j$ are nonexpansive for each $i,j.$ Select $r_n=\frac{2n}{2n+1}, \theta_n=0.8, \tau_n=0.7.$ It can easily be checked that all the conditions of Theorem \ref{the3} are satisfied. Now, we compute $T_r^{(F_1,\phi_1)}(x).$ We find $z\in C$ such that for all $y\in C$
		\begin{align}\label{Eq5.1}
			F_1(z&,y) + \phi_1(z,y) + \frac{1}{r}\langle y - z, z - x \rangle \geq 0\nonumber\\
			\Leftrightarrow & \langle \frac{z}{2}, y-z \rangle + \frac{1}{r}\langle y - z, z - x \rangle \geq 0\nonumber\\
			\Leftrightarrow & \frac{z}{3}(y-z) + \frac{1}{r}(y - z)(z - x)\geq 0\nonumber\\
			\Leftrightarrow & (y - z)[rz + 3(z-x)]\geq 0\nonumber\\
			\Leftrightarrow & (y - z)[(r+3)z -3x]\geq 0.
		\end{align}
		By Lemma \ref{lem3c}, we obtain
		\begin{equation*}
			T^{(F_1, \phi_1)}_r (x) = \left\{z \in C : F_1(z,y) + \phi_1(z,y) + \frac{1}{r} \langle y - z, z - x \rangle \ge 0, ~~ \forall~ y \in C \right\}, 
		\end{equation*}
($\forall~~x \in H_1$), is single-valued. Thus, from (\ref{Eq5.1}) we obtain $z=\frac{3x}{r+3}.$ This implies that $T^{(F_1, \phi_1)}_r (x) = \frac{3x}{r+3}.$  Similarly, we compute $T^{(F_2, \phi_2)}_r (v).$ We find $w\in Q$ such that for all $d\in Q$
		\begin{equation*}
			T^{(F_2, \phi_2)}_s (v) = \left\{ w \in Q : F_2 (w,d) + \phi_2(w,d) + \frac{1}{s} \langle d - w, w - v \rangle\ge 0,\quad \forall~ d \in Q \right\}.
		\end{equation*}
		By using similar approach as above, we obtain $w=\frac{4v}{s+4}.$ This implies that $T^{(F_2, \phi_2)}_s (v)=\frac{4v}{s+4}.$
		
	\noindent Here, we set the parameter $b$ on $\{\delta_{n,i}\}$ in \eqref{Eq5.1A} to be $b=3, v=t^2$ and we choose different initial values as follows:\\
		\noindent {\it Case I:} $x_0 = t^4, x_1 = t^2 + t^4 + t^6 + 3;$\\
		\noindent {\it Case II:} $x_0 = t^5,  x_1 = t^2 + t^5 + 2;$\\
		\noindent {\it Case III:} $x_0 = t^4, x_1 = t^3 + t^5 + t^7 + 2;$\\
		\noindent {\it Case IV:} $x_0 = t^5, x_1 = t + t^2 + 1.$\\\\
		\noindent We compare the performance of our Algorithm \ref{alg3a} with Algorithms \eqref{algdep}, \eqref{algsitth},  \eqref{algphu} and \eqref{xii}. The stopping criterion used for our computation is $||x_{n+1}-x_{n}||< 10^{-4}$. We plot the graphs of errors against the number of iterations in each case.  The numerical results are reported in Figure \ref{fig2} and Table \ref{tab2}.
	

\begin{table}[h]
		\caption{\bf Numerical results for Example \ref{EX2}}
		\label{tab2}
		\begin{tabular}{ |p{1.3cm} |p{2.5cm}| p{1.2cm}| p{1.2cm}| p{1.2cm}| p{1.2cm}|  p{1.2cm}|}
			\hline
			\noindent & \noindent & Alg. \eqref{algdep} & Alg. \eqref{algsitth} & Alg. \eqref{algphu} & App. \eqref{xii} & Alg. \ref{alg3a}\\
			\hline
			Case I & No. of Iter.  & 10 & 14 & 10 & 6 & 6\\
			&   CPU time (sec)  & 0.7297 & 0.7237 & 1.2541 & 0.2548 & 0.3256\\
			\hline
			Case II & No. of Iter. & 9 & 14 & 9 & 6 & 6\\
			&   CPU time (sec)  & 0.6743 & 0.7004 & 1.1791 & 0.2628 & 0.3091\\
			\hline
			Case III & No. of Iter. & 9 & 14 & 9 & 6 & 6\\
			&   CPU time (sec)  & 0.6507 & 0.6825 & 1.1474 & 0.2599 & 0.3087\\
			\hline
			Case IV & No. of Iter. & 9 & 13 & 8 & 6 & 6 \\
			&   CPU time (sec)  & 0.6353 & 0.6458 & 1.1130 & 0.2631 & 0.3166\\
			\hline

			
			
		\end{tabular}
	\end{table}
	
	
	\begin{figure}
		\begin{center}
			\includegraphics[height=4cm]{CaseA.pdf}	
			\includegraphics[height=4cm]{CaseB.pdf}\\
			\includegraphics[height=4cm]{CaseC.pdf}	
			\includegraphics[height=4cm]{CaseD.pdf}\\
		\end{center}
		\caption{ Top left: Case I; Top right: Case II; Bottom left:  Case III; Bottom right: Case IV.}\label{fig2}
	\end{figure}
	
		
	
	\end{example}	
	
	
	
	
	
	
	
	\section{Conclusion}\label{Sec7}
	\noindent In this article, we proposed a new modified inertial shrinking projection algorithm for finding common solution of split generalized equilibrium problem, monotone inclusion problem and fixed point problems for a countable family of strictly pseudo-contractive multivalued mappings. We established strong convergence result for the proposed method. We applied our results to study related optimization problems and presented  some numerical examples to demonstrate the efficiency of our proposed method in comparison with other existing methods. Our results extend and improve several existing results in this direction in the current literature.
	
\noindent {\bf Acknowlegment:} The first author acknowledges with thanks the bursary and financial support from Department of Science and Innovation and National Research Foundation, Republic of South Africa Center of Excellence in Mathematical and Statistical Sciences (DSI-NRF COE-MaSS) Doctoral Bursary. Opinions expressed and conclusions arrived are those of the authors and are not necessarily to be attributed to the CoE-MaSS.



\end{document}
